UNIVERSITY OF OKLAHOMA

GRADUATE COLLEGE

ASSESSING USER NEEDS AND MODEL ACCURACY OF SEASONAL CLIMATE FORECASTS FOR WINTER WHEAT PRODUCERS IN THE SOUTH-CENTRAL UNITED STATES

A DISSERTATION

SUBMITTED TO THE GRADUATE FACULTY

in partial fulfillment of the requirements for the

Degree of

DOCTOR OF PHILOSOPHY

By

TONI KLEMM Norman, Oklahoma 2018

ASSESSING USER NEEDS AND MODEL ACCURACY OF SEASONAL CLIMATE FORECASTS FOR WINTER WHEAT PRODUCERS IN THE SOUTH-CENTRAL UNITED STATES

A DISSERTATION APPROVED FOR THE DEPARTMENT OF GEOGRAPHY AND ENVIRONMENTAL SUSTAINABILITY

 $\mathbf{B}\mathbf{Y}$

Dr. Renee A. McPherson, Chair

Dr. Elinor Martin

Dr. Travis Gliedt

Dr. Jennifer Koch

Dr. Jean Steiner

© Copyright by TONI KLEMM 2018 All Rights Reserved. Für meine Eltern

Acknowledgements

Had someone told me in 2012 how much blood, sweat, and tears go into getting a Ph.D., I don't think I would have ever started. But luckily, no one told me. While the last five and a half years were a rollercoaster ride of joy and sadness, anxiety and relief, happiness and frustration, I would not want to trade it for anything. I learned to live in a new culture, I traveled to fascinating places and met fascinating people, I made friends and found my true purpose, and in a few months I will marry the woman of my dreams and will stay in the U.S. probably a bit longer than I originally intended.

Although this dissertation has my name on the front, none of this work would have been possible without the generosity of many people who contributed their time and effort so I could succeed. There is no way I can thank everyone, so I will cherry-pick the most dear to my heart and accept that many other will inevitably be left out. I thank my graduate committee, Drs. Jean Steiner, Travis Gliedt, Elinor Martin, and Jennifer Koch, for their enduring support, their insightful advice, and their scrutiny of my work, which ultimately made it much better. I am especially thankful for my dissertation advisor, committee chair, mentor, and amazing role model, Dr. Renee McPherson. Renee trusted my abilities long before I did. She always struck the perfect balance of helping me and letting me figure things out. I admire her patience, humility, and appreciation of others, and observing her leadership made me a better researcher and fundamentally shaped my idea of how to act and lead in a professional environment.

My appreciation goes out to OU's Department of Geography and Environmental Sustainability, to the College of Atmospheric and Geographic Sciences, the Oklahoma EPSCoR Program, and the South Central Climate Science Center for funding my research and for providing generous travel support to attend conferences and workshops, for giving me opportunities to speak and present my research, to advise young students, and to be an advocate for careers in geography and atmospheric sciences.

I tip my hat to my coworkers at the South Central Climate Science Center for their technical expertise, especially to Drs. Esther Mullens, Derek Rosendahl, Mike Langston, Kim Winton, and Adrienne Wootten, to Mark Stacy and Tyler Pearson, and to Terri Sarsycki, and am hugely thankful for their professional guidance, for teaching me many new skills, and most of all for their friendship. Working with all of you was huge fun and a privilege that will remember a long time. I consider all of you friends.

It is impossible to overstate how much my friends and family supported me. To my parents, Carmen and Reiner, I am grateful for passing their values of care and thoroughness on to me, for giving me world interest, and for supporting every endeavor I pursued. Words cannot express my gratitude to you. To my fiancée, Stephanie, I am endlessly thankful for caring for me, for her encouragement, for listening to me complain about struggles and helping me resolve them, and for pulling me away from work when it was necessary. You motivated me to keep going. Thanks to all my friends in Norman. Time spent with you made all the difference, and I hate to say goodbye. And to my friends in Germany, we'll catch up soon. Sorry for being slow in responding lately. I *will* get to your emails, letters, and postcards, I promise.

Finally, thanks to God. He kept me going during what was the by far most challenging and exhausting time in my life thus far. Everything I accomplished was made possible through his grace, his mercy, and his provision.

As the apostle Paul wrote:

"Now to him who is able to do immeasurably more than all we ask or imagine, according to his power that is at work within us, to him be glory in the church and in Christ Jesus throughout all generations, for ever and ever! Amen." (Ephesians 3.20-21, NIV)

Acknowledgements	iv
List of Tables	ix
List of Figures	x
Abstract	xvi
Chapter 1 - Introduction	1
Chapter 2 - The Development of Seasonal Climate Forecasti for Agricultural Producers	ng 6
1 Introduction 1.1 Historical Overview 1.2 Recent Efforts in Seamless and Extreme Events Forecasting	8 8 10
 2 Seasonal forecasts and their application in agriculture 2.1 Seasonal tercile and POE forecasts 2.2 Multimodel Ensemble Seasonal Forecasts 2.3 Applications and benefits for agriculture 	12 12 16 18
 3 Skill and utility as measures of forecast value	21 21 24
4 Seasonal climate forecasts in the context of other farm decision factors	26
5 Forecast demands and forecast communication 5.1 Forecast demands from producers and agricultural advisors 5.2 Communication Challenges	28 29 40
6 End-to-End Concept	43
7. Conclusion	45
8. Acknowledgements	46
9. Conflict of Interest Statement	46

Chapter 3 - Assessing Decision Timing and Seasonal Climate Forecast Needs of Winter Wheat Producers in the South-Central United States 47

1. Introduction	
2. Methods	
2.1 Why winter wheat?	
2.2 Survey population and distribution	
2.3 Survey design	54
3. Results	56

3.1 General survey statistics	
3.2 Decision timing	
3.3 Weather and climate threats	62
3.4 Forecast preferences	63
4. Discussion and Conclusion	66
5. Acknowledgements	69

Chapter 4 - Assessing Model Accuracy of Seasonal Climate Forecasts for Winter Wheat Producers in the South-Central United States

United States	71
1. Introduction	73
2. Study area, data, and methods	74
2.1 Study area climatology	75
2.2 Observational and model datasets	75
2.3 Survey data input	77
2.4 Data preparation and processing	79
3. Results	
3.1 Total monthly precipitation	
3.2 Average monthly air temperature	90
3.3 Number of days per month with extreme precipitation	93
3.4 Number of dry days per month	107
4. Summary and discussion	115
5. Conclusion	117
6. Acknowledgements	118
7. Conflict of interest statement	118
Chapter 5 - Summary and Conclusion	119
1. Summary	
2. Conclusion	

Bibliography

List of Tables

Tab. 1: Agricultural decisions connected to different types of meteorological variables.Source: Frisvold and Murugesan (2012)
Tab. 2: Summer and winter crop stage calendar for the US. Source: USDA (2015b)33
Tab. 3: Corn growers' needs for forecast information at specific times of the crop year. Source: (2011)
Tab. 4: Farm management decisions related to different forecasts lead-times. Source: Hudson (1972)
Tab. 5: Farm management decisions related to seasonal climate forecasts. * indicates decision that are also influenced by forecasts of shorter time scales, such as extended weather forecasts or short-term weather forecasts. Source: Excerpt from Mavi and Tupper (2004)
Tab. 6: Number of survey responses by state and number of responses without location information. 54
Tab. 7: Date of peak planning times and average decision time span, calculated by creating the average of the last month extension agents selected for every decision and subtracting it from the average first month agents selected for every decision
Tab. 8: Ranking for forecast elements based on monthly decision-timing and forecast preferences for each decision
Tab. 9: Description of the observational and model datasets used in the study75
Tab. 10: Definitions or precipitation thresholds used in the study80
Tab. 11: Precipitation datasets and extremes indices used. 98

List of Figures

Fig. 5: Climate-based decision cycle for corn. The outer calendar identifies the time of year management decisions are made. The inner calendar depicts the soil or crop impact, and the label on the arrow identifies the weather or soil conditions relevant to the impact. Length of the arrow gives the lead-time of climate forecasts that links the specific agricultural decisions to soil or crop impacts. Source: Takle et al. (2014)...... 34

Fig. 6: Schema of the end-to-end concept. Actionable outcomes require interaction and feedback from societal, biological, and physical elements. Source: Goddard et al. (2001)

Fig. 9: This matrix shows when winter wheat growers plan for certain agricultural practices. The calendar months are listed on the x-axis, labeled at the bottom; practices are listed on the y-axis, labeled on the right. The size of the bubble represents how often a given practice was selected for that given month by extension agents, with larger

Fig. 11: This matrix shows what management practice decisions are impacted by which weather and climate threats. The practices are listed on the x-axis, labeled at the bottom; threats are listed on the y-axis, labeled on the right. The size of the bubble represents how often a given practice was selected for that given month by extension agents, with larger bubbles representing more responses. The ranking of threats is based on the total number of times they were associated with a practice. It shows that the most impactful threat is drought, followed by extreme rainfall. The matrix also highlights that some threats impact some practices more than others, e.g., hail impacts harvest time more than any other listed practice.

Fig. 14: Biases at lead zero for (A) monthly total precipitation and (B) monthly average temperature by forecast month. Note that the y axis in (A) is logarithmic. Errors above one (zero) indicate overestimation of the model for precipitation (temperature); errors

Fig. 18: Time series showing the differences of absolute model and absolute persistence forecast errors from January to December from lead zero to 11. The thick line represents the average, thin lines the minima and maxima across the study area for a respective lead-time. Values above (below) zero indicate smaller absolute persistence (model) error.

Fig. 36: The percentage of study area with smaller absolute persistence error (green) and smaller absolute model error (red) for forecasts at lead zero, four, seven, and 11. 113

Fig. 37: Percentage of the numbers of dry days/month as predicted by persistence and model forecast. The dashed red line shows the average persistence error, the blue shaded

solid lines and the grey shaded dotted lines show the model error for leads zero to five and six to 11 respectively. The black dash-dotted line shows the model lead zero to 11 average. Note: The underlying absolute numbers are based on averages of the raw data (the same data that was used in Fig. 27 and Fig. 28); therefore, the percentages here do not match the absolute forecast errors of the respective month (and lead-time). 114

Abstract

The research presented in this dissertation highlights ways in which seasonal climate forecasts can be tailored to better serve the needs of winter wheat producers in the southcentral United States (U.S.) and presumably in other regions, and address previously raised criticism of these forecasts by the agricultural community. It applied a collaborative, interdisciplinary approach and conducted a quantitative online survey of agricultural advisors to determine decision timing and seasonal forecast needs in winter wheat production in Texas, Oklahoma, Kansas, and Colorado. These results were used to create a ranking that showed forecast elements most requested are related to precipitation and consist of information directly modeled, such as average total precipitation or average temperature, and data derived from such information, such as connective days without precipitation or chances of extreme precipitation. A subsequent analysis used this ranking to conduct a error comparison of a high-resolution seasonal climate model and a persistence forecast derived from observational data. Survey results show that current seasonal climate forecast omit several forecast elements important in winter wheat production, which current seasonal forecast models are capable of producing, such as the number of consecutive days without rainfall or the chances for extreme rainfall. Results of the seasonal forecast analysis showed that the seasonal climate forecast model used had a greater absolute error than a seasonal persistence forecast for all forecast elements across most of the study region and most of the year.

Results contribute significantly to the current body of knowledge in tailored seasonal climate forecasting and highlight the fact that both model and persistence forecast can be more accurate, depending on the forecast element, time of year, geographic location, and

xvi

lead-time, and that in some cases, both model and persistence forecasts may be very inaccurate.

Chapter 1 - Introduction

Agriculture is a highly weather- and climate-dependent industry. Unseasonal warm or cold periods and extreme events like drought or extreme rainfall can severely impact agricultural production and have serious societal impacts as a result of food shortage and/or food price fluctuations. As seen in recent years in the United States (U.S.), East Africa, and Syria, severe drought and heat can lead to severe crop damage and crop failure (NRDC 2013), affect regional economies (Cargill 2015), cause spikes in insurance payments (U.S. Department of Agriculture 2017), affect food security and food prices (USDA ERS 2013), and contribute to famine and violent conflicts, such as recently in East Africa (Marthews et al. 2015) and Syria (Gleick 2014).

Seasonal climate forecasts, forecasts with lead-times of one to 12 months issued monthly as one- or three-month averages or totals, can help mitigate and reduce these negative impacts. Proactive farm decision making that adapts to uncertain weather and climate conditions has been shown to stabilize crop yields, reduce revenue fluctuations, provide a more stable income over time compared to conservative, business-as-usual practices, and even increase yields by taking advantage of more favorable conditions (Meinke et al. 2003, Meinke and Stone 2005, Nicholls 1980). Seasonal climate forecasts, forecasts with monthly or seasonal averages and lead-times of one to 12 months, have been issues for the contiguous U.S. since 1946 (O'Lenic et al. 2008) and have been used by agricultural producers for decades, however not without criticism (Changnon, Sonka and Hofing 1988, Sonka, Hofing and Changnon 1992). Among other factors, producers critiqued the lack of relevant information and the disconnect between forecasters and users (Schneider and Wiener 2009), what informally is called the "loading dock approach" by the National Weather Service (NWS)(Cash, Borck and Patt 2006), issuing forecasts without tracking who uses them and without feedback or initial input from users. In response, Cash et al. (2006) proposed science co-production across disciplinary boundaries, scales, and knowledges, to produce "to create information that is salient, credible, and legitimate to multiple audiences" (Cash et al. 2006), a task that in the past, due to its cross-disciplinary, applied nature seemed particularly suited for geographers (Moser 2010).

Over the past decades, advances in the understanding of atmospheric, land, and ocean processes and their interaction have led to substantial skill improvements in seasonal climate forecasting in many parts of the worlds (Delworth et al. 2006, Delworth et al. 2012, Goddard et al. 2001, Jia et al. 2015, Kirtman and Min 2009, Kirtman et al. 2014). In order to translate this progress (and many other advances) into improved decision making and societal applications in the U.S., several programs, such as the National Oceanic and Atmospheric Administration (NOAA) Regional Integrated Sciences and Assessment (RISA) program (Lemos and Morehouse 2005), the Department of the Interior Climate Science Centers (DOI 2009), and the U.S. Department of Agriculture Regional Climate Hubs (Allen and Stephens 2016) were initiated between the late 1990s and early 2010s. Major goals of these programs were to provide climate-related decision support by bringing together scientists and stakeholders and building lasting relationships across disciplines, moving from the loading dock approach towards "deliberate coproduction, which involves explicitly planning coproduction into research processes and applying the best practices in collaborative research to achieve usable science" (Meadow et al. 2015). The result of these efforts are end-to-end approaches, more holistic and collaborative approaches to science production, which are aimed at

considering the entire forecast production cycle, from developing models to tailoring information for specific decisions in collaboration with users (Bales, Liverman and Morehouse 2004, Lyon et al. 2014, Roncoli et al. 2009, Shafiee-Jood et al. 2014).

Despite efforts to derive applications from long-term climate information, recent research often focused solely on the assessment of user needs, in particular with regards to the agricultural community (Takle et al. 2014, Schneider and Wiener 2009, Cabrera et al. 2006), while intra-seasonal to inter-annual forecasts are being improved with the potential of seasonal climate forecast products tailored in ways requested by users. The research presented in the following chapters embraces the collaborative, interdisciplinary efforts outlined in the previous paragraph and intended to move one step further towards operational tailored forecasts. The work presented here is built on the idea of combining current knowledge in seasonal forecast production and user needs closer together.

This research was intended to focus on one single crop and explore ways in which seasonal climate forecasts can be tailored to help producers improve their decision making. The crop chosen for this research was winter wheat, a strain of wheat and the largest crop by acreage in the south-central U.S., grown on 21.1 million acres (in 2016), twice the area of the second largest crop, corn (Han et al. 2012). Winter wheat also contributes about 71 percent to the total U.S. harvest of wheat, which itself is the third largest U.S. field crop behind corn and soybean (USDA 2012). With this, the goal of this work was to answer two research questions:

1. How can seasonal climate forecasts be tailored to serve the needs of winter wheat growers in the south-central United States?

2. Can existing seasonal forecast models provide meteorological variables as requested by winter wheat farmers with better skill than a persistence forecast?

The following three chapters are three publications from this research, which have been published, are currently under review, or are in draft. Chapter two gives a historic review of the development of seasonal climate forecasts for agricultural producers, predominantly since the year 2000. Chapter three presents a survey study conducted in 2016 in Texas, Oklahoma, Kansas, and Colorado, to assess decision timing of major farm practices in winter wheat production in the south-central U.S. and to determine seasonal forecast needs of winter wheat producers. Chapter four presents a forecast model analysis that used results from chapter three to feed a statistical comparison of a high-resolution seasonal forecast model and a persistence forecast for the survey study domain, determining whether a model forecast or a persistence forecast are more accurate in providing forecasts as requested by winter wheat producers in the study area. Finally, Chapter five summarizes all results and draws an overall conclusion.

Chapter 2 - The Development of Seasonal Climate Forecasting for Agricultural Producers

Published as:

Klemm, Toni & Renee A. McPherson (2017): The development of seasonal climate forecasting for agricultural producers. Agricultural and Forest Meteorology, 232, 384-399.

Abstract

This review summarizes advances in seasonal climate forecasting with a focus on agriculture, predominantly since the year 2000. The main research methods used were keyword searches in publisher-unaffiliated databases such as Web of Knowledge and in publication libraries of institutions known for their interdisciplinary work in climate forecasting and agriculture. Crop and livestock producers use seasonal climate forecasts for management decisions such as planting and harvest timing, field fertilization, or grazing. Agricultural users have often criticized lack of forecast skill and usability as well as a lack of understanding of user needs among forecast developers. Recently, interdisciplinary studies started exploring agricultural decision-making and integrating social science and climate science in order to improve the value of seasonal forecasts. Producer requests include direct and derived forecast products, such as total rainfall and consecutive dry days, information on uncertainty, and comparisons to previous years. The review explores single-model and ensemble forecasts, describes different measures of forecast value, and highlights economic and other agricultural decision factors besides weather and climate. It also examines seasonal climate forecasts from an agricultural perspective, explores communication challenges and how to overcome them, and delves into end-to-end forecast concepts that span forecast production to forecast application by end users.

1 Introduction

Seasonal climate forecasts, forecasts with lead-times of one to 12 months issued monthly as one- or three-month averages or totals, are used by crop producers for management decisions, for example planting and harvest timing, field fertilization, or water management, among others. To review the literature, the main methods used were keyword searches in publisher-unaffiliated databases such as Web of Knowledge and in publication libraries of institutions known for their interdisciplinary work in climate forecasting and agriculture, such as the Southeast Climate Consortium and the International Research Institute for Climate and Society. Section 1 outlines a brief history on the topic. Section 2 explores single-model and ensemble forecasts. Section 3 reviews measures of skill and utility to assess the value of seasonal climate forecasts. Section 4 puts seasonal climate forecasts in context with other agricultural decision factors. Section 5 looks at tailored seasonal climate forecasts and seasonal climate forecasts from an agricultural perspective, while exploring communication challenges between forecast provider and forecast user, and highlights how agricultural decision-making differs depending on the crop type and planning horizon. Lastly, section 6 explores end-to-end concepts.

1.1 Historical Overview

The benefits of weather and climate monitoring for agricultural purposes have been recognized for several centuries. Thomas Jefferson, third president of the United States (U.S.) (1801-1809), planned farming operations based on local climate conditions (Changnon 2007). From 1776 to 1816, he kept an almost continuous record of daily weather conditions (Fiebrich 2009). Long-range weather forecasts date back to at least 1793, the first publication of the Farmer's Almanac, an annual general interest magazine

containing descriptive weather forecasts for the coming calendar year (Hale 1991). Forecasts in the Farmer's Almanac were based on reoccurring weather patterns and proxy data, such as the thickness of the skin of onions, width of the stripes on the woolly caterpillar, or moon phases; however, the exact forecasting method remains unpublished (Hale 1991). A survey of agricultural advisors to midwestern U.S. corn growers found that 95 percent of participants knew of the Farmer's Almanac, but only 18 percent indicated using it (Prokopy et al. 2013).

Native American Tribes were recognized for their long-term forecasts, as a 1950-to-1952 letter by Senator Robert S. Kerr (D-Oklahoma) reveals. Despite having access to U.S. Weather Bureau (predecessor of the National Weather Service, NWS) forecasts, Kerr wanted to "know what some of the Indians in the various sections of the nation think about our coming winter probabilities" (Peppler 2010 200). Much like the Farmer's Almanac, Tribes appeared to base their predictions on natural phenomena, such as the thickness of corn shuck and how many spider webs were in the air and in trees (Peppler 2010).

Other authors reviewed the then-current status of long-term weather predictions and seasonal forecasting. For example, Namias (1968) summarized historical developments and important literature on long-range forecasting, while Nicholls (1980) gave an historical overview of seasonal forecasting methods, such as analogs, teleconnections, cosmic cycles, time series, and early numerical modeling. Goddard et al. (Goddard et al. 2001) reviewed predictability and prediction of seasonal to inter-annual forecasts, including statistical and dynamical forecast methods and forecast performance, and Goddard et al. (2012) compared seasonal forecasting to decadal forecasting. None of

these summaries, however, focused on the usefulness of seasonal forecasts to agricultural producers.

In 2006, the World Meteorological Organization (WMO) set production and verification standards for seasonal climate forecasts that are currently followed by 12 national and multinational forecast centers on five continents, so-called Global Producing Centers (GPC) (WMO 2015). According to these standards, nations must forecast air temperature (2-m height), precipitation, sea-surface temperature, mean sea-level pressure, 500hPa height, and 850hPa temperature, issued at least every three months with minimum lead-times between zero and four months (WMO 2015).

1.2 Recent Efforts in Seamless and Extreme Events Forecasting

With improvements in supercomputing and advances in understanding of physical processes both in the atmosphere as well as between atmosphere, oceans, and land surfaces, modelers can apply techniques from numerical weather prediction to numerical climate prediction. "[S]cientifically, predicting weather at shorter ranges, or its various statistics at longer time ranges, is based on the same laws of physics" as forecasting for longer time scales (Toth, Peña and Vintzileos 2007 1427). This transition to longer time scales is desirable in an effort to provide seamless, skillful forecasts from hourly to seasonal time scales (Toth et al. 2007). In practice, several programs have been initiated to work towards seamless forecasting. From 2006 to 2016, THORPEX ("THe Observing system Research and Predictability EXperiment"), a WMO program supervised by the U.S. National Oceanic and Atmospheric Administration (NOAA), developed procedures and devised research priorities to extend the 7- to 10-day limit of numerical weather prediction out to 14 days. THORPEX was also intended to develop intraseasonal forecasts of up to

60 days lead-time (Toth et al. 2007). DEMETER ("Development of a European Multimodel Ensemble system for seasonal to inTER-annual prediction"), a European project from 2000 to 2003, used downscaling techniques to produce high-resolution global seasonal climate forecasts from an ensemble of European seasonal forecast models (three or seven models, depending on the length of the hindcast period) as input for other prediction models, for example for crop yields or the distribution of diseases like malaria (Palmer et al. 2004).

DEMETER's successor, ENSEMBLES ("ENSEMBLE-based predictions of climate changes and their impacts"), ran from 2004 to 2009 and improved DEMETER's seasonal forecast performance (for example in the northern midlatitudes, and in lead-times of 4 to 6 months) by using an ensemble of nine updated forecast models (Weisheimer et al. 2009, Alessandri et al. 2011). From 2012 to 2016, EUPORIAS ("EUropean Provision of Regional Impact Assessment on a Seasonal-to-decadal timescale"), another European, multi-institutional program, developed probabilistic forecasts of high-risk events for Europe and parts of Africa for seasonal to decadal timescales (Hewitt, Buontempo and Newton 2013). EUPORIAS facilitated 24 national and multinational European forecast centers with expertise in seasonal forecasting, impacts assessment, and new media communication, as well as climate-sensitive industries such as agriculture, energy, and tourism, to create decision-support tools for these industries and to increase their competitiveness (Hewitt et al. 2013).

In 2013, the S2S (Sub-seasonal to Seasonal) prediction project was initiated by the World Weather Research Programme (WWRP), World Climate Research Programme (WCRP), and THORPEX as a five-year project to foster international research collaboration and to fill the forecast gap between medium-range weather forecasts (up to 2 weeks) and seasonal forecasts of 3 to 6 months (Robertson et al. 2015). S2S objectives, science plans, and descriptions of individual sub-projects can be viewed on <u>www.s2sprediction.net</u>. Ringler et al. (2008) and Hoskins (2013) point out that forecasts on different time-scales, from days to centuries, can be affected by distinct phenomena and components of the natural world that need to be researched and taken into account for skillful forecasting, such as fronts and convective systems, ocean circulation, or vegetation cover. Seamless forecast model development included the Model for Prediction across Scales (MPAS) (Ringler et al. 2008) developed at the National Centers for Atmospheric Research (NCAR) and ICON ("ICOsahedral Non-hydrostatic general circulation model"), developed by the German Weather Service and the Max Planck Institute for Meteorology (Zängl et al. 2014). Both MPAS and ICON could also benefit seasonal forecasting, for example for agriculture. In addition to government products, many long-range forecasts are being produced by commercial providers, as overviewed by Hartmann et al. (2002).

2 Seasonal forecasts and their application in agriculture

After highlighting scientific achievements towards seasonal and seamless forecasting, the following sections will focus on the development of two seasonal climate forecast efforts carried out by the U.S. Climate Prediction Center and illustrate how these products can be applied to agricultural decision-making.

2.1 Seasonal tercile and POE forecasts

The U.S. Climate Prediction Center (CPC) has been issuing long-range forecasts for the contiguous U.S. since 1946 (Kerr 2008), for example for crop producers or natural gas suppliers (Kerr 1989 30). Until 1981, these forecasts had no lead-time (i.e., the forecast



Fig. 1: CPC three-month temperature forecast for June/July/August 2015, issued on May 21, 2015 (0.5 months lead-time). Source NOAA CPC.

started with the issue date), and forecasts were three-month aggregated and based on three probabilistic categories (above, below, and near the long-term average) (O'Lenic et al. 2008). Better understanding of El Niño in the 1980s and 1990s (e.g., Rasmusson and Carpenter 1982) led to advances in the skill of seasonal forecasts such that, in 1995, CPC changed the forecast format into a series of 13 consecutive, overlapping 3-month periods, issued every month, starting at lead-times of 0.5 months (Fig. 1: CPC three-month temperature forecast for June/July/August 2015, issued on May 21, 2015 (0.5 months lead-time). Source NOAA CPC.), and ending at 12.5 months (O'Lenic et al. 2008). The 0.5 minimum lead-time was to avoid redundancy with daily weather forecasts (van den Dool 1994) while 12.5 months was long enough to cover an entire crop year while not significantly losing forecast skill (van den Dool 1994). Agricultural producers and related industries embraced these improvements and started using seasonal forecasts more frequently (Changnon 2004). In 2006, CPC started using a new forecast model, the Climate Forecast System version 2 (CFSv2) which has improved predictive skills over its predecessor (O'Lenic et al. 2008, Peng, Barnston and Kumar 2013, Saha et al. 2014).

There are two major ways in which CPC's probabilistic seasonal forecasts are displayed: tercile maps and probability of exceedance (POE) graphs. Tercile maps (Fig. 1: CPC three-month temperature forecast for June/July/August 2015, issued on May 21, 2015 (0.5 months lead-time). Source NOAA CPC.) indicate which regions will most likely experience above-, below-, or near-normal conditions for temperature and precipitation. Although terciles are the most commonly used format, they have a number of disadvantages. They do not contain much spatial detail, and they have been criticized by users for not being communicated in an "obvious, user-friendly format" (Barnston, He and Unger 2000 1272) and for not having enough skill to be considered in decision-making (Barnston et al. 2000). They also lack information on forecast uncertainty (Barnston et al. 2000) and cannot quantify the amount of deviation in temperature or precipitation from normal, reducing their utility for agricultural producers (Garbrecht et al. 2010).

Barnston et al. (2000) argue that seasonal climate forecasts would be considered more seriously by users "if more plentiful and detailed information were offered both in the forecasts themselves and in descriptions of their expected accuracy" (Barnston et al. 2000



Fig. 2: Probability of Exceedance (POE) curve for average temperature forecast in central Oklahoma (forecast division 53) for May to July 2015, issued April 16, 2015. Source: NOAA CPC

1272). POE graphs (Fig. 2) are less common but provide more information than tercile maps. They show the probability (on the y-axis) that a given temperature (on the x-axis) will be exceeded during a particular time period (e.g., May-July 2015), both for that period (black line in Fig. 2) and for the historic average (red line in Fig. 2). POE graphs can be viewed as "a backward cumulative probability density function" (Barnston et al. 2000 1273) because the probability of exceeding a certain temperature threshold decreases with increasing temperature. CPC calculates POE graphs for each of the 102 NWS forecast divisions in the U.S. (see also Fig. 4), and thereby provide more spatial detail than their tercile maps. POE graphs also provide information on forecast uncertainty, a feature often requested by agricultural producers. Seasonal climate forecasts in POE graph format have been used to create a forecast precipitation index for water managers in the

southeastern U.S., resulting in potential benefits for the state of Georgia estimated between \$30 million and \$350 million per year (Steinemann 2006). These products have also been used to estimate surface runoff in the U.S. (Garbrecht, Schneider and Van Liew 2006), and to estimate the utility of seasonal forecasts for U.S. agricultural producers (Schneider and Garbrecht 2003a, Schneider and Garbrecht 2003b, Schneider and Garbrecht 2006).

2.2 Multimodel Ensemble Seasonal Forecasts

Increased multinational collaboration and nearly four decades of research into the origins of seasonal predictability brought two major advances in seasonal forecasting in the early 2000s: inclusion of quantitative information about uncertainty, and recognition that multi-model ensembles are a viable option to reduce forecast uncertainty (over single model approaches as used by CPC), both of which help serve end users with better decision-support (Kirtman et al. 2014). An implementation of these advances is exemplified in the North American Multimodel Ensemble (NMME), which went operational in 2012. It creates global seasonal forecasts by averaging forecasts from several individual seasonal forecast models, each of which is run with a range of different initial conditions.

Averages of ensemble forecasts are considered more skillful than single-model forecast averages because multiple models can average out errors of individual models (Stockdale et al. 2010, DelSole and Tippett 2014). Tests with NMME in Kirtman et al. (2014) also came to this conclusion, in particular with respect to comparing sea and land-surface temperature and precipitation forecasts between NMME and CFSv2 from 1982 to 2009. Furthermore, Infanti and Kirtman (2014) found low error and high skill in NMME temperature and precipitation forecasts for the southeastern U.S. during the El Niñoaffected winter of 2006/2007.

NMME produces forecasts in one- and three-month aggregates, with lead-times of one to seven months for mean temperature and total precipitation anomalies (Kirtman et al. 2014). Fig. 3 exemplifies an NMME forecast for North America for October 2015, issued one month earlier. In 2015, forecasts for additional variables became available: global geopotential height at 200hPa atmospheric pressure, global and U.S. maximum and minimum temperature, U.S. soil moisture, and U.S. runoff. The spatial resolution of the forecasts is 1° latitude by 1° longitude (Infanti and Kirtman 2014). To determine the forecast skill of the ensemble, hindcasts were compared against observations using anomaly correlation, root-mean square error, reliability, and the ranked-probability skill



Fig. 3: NMME ensemble forecast for precipitation deviation from normal for October 2015, issued in September 2015. Source: NOAA CPC

score (Kirtman et al. 2014). As of 2016, seasonal forecasts and forecast skill are available in map and table formats at http://www.cpc.ncep.noaa.gov/products/NMME/.

2.3 Applications and benefits for agriculture

Agricultural producers and related industry users (e.g., seed manufacturers, fertilizer producers, cooperative extension, weather insurance) have long had interest in climate forecasts for decision support (Sonka et al. 1992, Boulanger and Penalba 2010, Frisvold and Murugesan 2012). Crop and livestock producers use weather and climate data for irrigation, planting, and harvest timing, selection of crop type and/or crop variety, decisions related to grazing, moving and selling of livestock, and decisions related to crop storage and purchase of crop insurance (Frisvold and Murugesan 2012). Yet, producers also remain skeptical of seasonal climate forecasts (Changnon et al. 1988) because of individual negative experiences in the past (Hu et al. 2006), for example. Agribusinesses use climate forecasts to develop new crop varieties (seed producers), to scout for locations for new plant sites (food processors), and to schedule production of fungicides and pesticides (agrochemical companies) (Changnon et al. 1988).

Although agribusinesses in the U.S. use climate forecasts, they also state that forecast accuracy and insufficient prediction lead-time were major impediments for using them for specific decisions (Sonka et al. 1992) which is why most of these applications were mainly restricted to historical climate records, rather than actual forecasts (Changnon et al. 1988). Year-to-date information (e.g., calendar year, crop year, or water year to the current date) and seasonal forecasts were found potentially useful by users and were valued higher than historical records (Changnon et al. 1988, Haigh et al. 2015), suggesting that agribusinesses would incorporate forecasts if they provided economic benefits
through increased reliability. Indeed, between 1981 to 2002, the value and usage of climate predictions for agribusinesses grew (Changnon 2004). This increase largely resulted from skill improvements in CPC's seasonal forecasts, more private firms that provided climate services based on these improved forecasts, new corporate orientations within seed producers and food processing companies, such as geographic diversification which allowed for hedging and lowered their overall weather- and climate-related risks, as well as increased competition and economic pressure, better understanding of climate predictions, improved information access, and more timely forecasts (Sonka et al. 1992, Changnon 2004, Templeton et al. 2014).

Decision experiments related to hedging weather risks show, forecast accuracy levels of only 50% had "considerable financial value" (Changnon 2004 611), and forecast accuracy of 65% or more "offered quite sizable corporate benefits" (Changnon 2004 611). Crane et al. (2010), meanwhile, describe risk management of Georgia family farmers as a mix of "planning and performance" rather than risking losses due to short-term adjustments: "The rationale for this approach is that consistency eventually pays off and that, in the long run, it is safer than trying to adjust cropping patterns seasonally to maximize short-term gain" (Crane et al. 2010).

Seasonal climate forecasts are also used as input for numerical crop models to produce or improve crop yield estimations. Mishra et al. (2008), for example, incorporated seasonal rainfall forecasts into the System of Agro-climatological Regional Risk Analysis version H (SARRA-H) crop model and improved sorghum yield predictions in Burkina Faso (for more information about SARRA-H, see Bontkes and Wopereis 2003). Roel and Baethgen (2007) tested warm, neutral, and cold El Niño Southern Oscillation (ENSO) phases in a crop simulation model for rice yield prediction in Uruguay for respective El Niño, La Niña, and neutral years. Although their crop model generally underestimated productivity, results from three different simulations had high correlations (r = 0.78 to 0.90; p = 0.0001) to observed yields (Roel and Baethegan 2007). Zinyengere et al. (2011) used five different ENSO phases (positive, neutral, negative, rising, and falling) to feed Rainman, a climate analysis tool, to produce monthly climate forecasts for these ENSO categories for a region in Zimbabwe. The resulting output was then used as input for a crop model, AquaCrop, to estimate maize yields and to serve as a starting point for a maize production decision-support tool.

Although crop models can help estimate yields using operational seasonal climate forecasts, steps need to be taken to merge the spatial and temporal scales in which crop models and climate models operate (Hansen and Indeje 2004). Hansen and Indeje (2004) point out that crop models usually operate on field-scale and daily resolution, while operational seasonal climate forecasts are often aggregated over three-month periods, to reduce noise and to increase forecast skill, and have a spatial resolution "on the order of 10,000 km²" (Hansen and Indeje 2004 144), multiple times larger than what is suitable for crop models (see Flato et al. 2013 854-866 for spatial resolutions of different climate models). To bridge those differences, Hansen and Indeje (2004) discuss two pathways: (1) using daily-resolution weather input directly from climate models or from stochastic weather generators, and (2) applying crop yield models that operate on the basis of climatic predictor variables, such as ENSO. Takle et al. (2014) also argue that higher spatial resolution is required to allow management decisions on a field-scale. "Decisions on crop and cultivar selection, tillage and conservation practices, fertilizer and chemical application, and planting and harvesting options require climate information that, ideally, is at the field scale" (Takle et al. 2014 4).

3 Skill and utility as measures of forecast value

In this section, forecast value as a function of forecast skill, predictability, and effectiveness of seasonal climate forecasts will be explored.

3.1 Forecast skill

By definition, climate models are a simplification of the earth's climate system and calculate atmospheric, land surface, and oceanic processes in a simplified way (NRC 2010). As a result, model predictions form an incomplete and imprecise picture of atmospheric processes, and they can also deviate substantially from each other because the research groups that assembled them used different algorithms, numerical techniques, and observational data for model initialization, calibration, and validation. Climate models and climate forecasts also have inherent errors because of the limitations of point measurement representing an entire area, measurement errors, and limited computing resources for processing at higher resolutions and with more complex physics (NRC 2010, IPCC 2013). Another skill-limiting factor is the inherent chaos of the weather system in which minute changes, undetectable by measuring devices and indescribable in equations, can have major effects on the development of future conditions (Lorenz 1969, Slingo and Palmer 2011). Skill scores can quantify these model errors, for example, by comparing a number of years of observations against a retrospective forecast for the same time period produced by the climate model or climate model ensemble of interest (Wilks 2011). By using hindcast predictions, one can assess how many of the ensemble realizations made "correct" predictions (i.e., how many were within a defined margin of error around the actual observation) and the magnitude of the total difference between forecast and observation (Richardson 2001, Kirtman et al. 2014). Skill information is an important metric to assess and compare in terms of accuracy of forecasts and climate

models; and knowledge about past forecast skill is valuable for crop and livestock producers in assessing and managing forecast uncertainty (Hansen 2002 319). More (less) skill may reduce (increase) the need to diversify grain crops or even change what livestock species are produced (Stern and Easterling 1999 64).

In many areas of the world, prediction skill and predictability of seasonal climate are determined by two factors: (1) the strength of dominant climate signals, like ENSO, in the region of interest, and (2) predictability of the ENSO signal itself, which reoccurs quasiperiodical (Latif et al. 1998). Via teleconnections, climate signals like ENSO determine weather and climate in many regions, such as Uruguay (Roel and Baethegan 2007), Paraguay (Ramirez-Rodrigues et al. 2014), Argentina (Jones et al. 2000), Mexico (Adams et al. 2003), the southeastern U.S. (Hansen, Hodges and Jones 1998, Jones et al. 2000, Hansen et al. 2001), or the U.S. Gulf Coast (Polade et al. 2013). Skill in seasonal climate forecasts in the U.S. is strongest for winter, weakest in summer, with spring and fall in between (van den Dool 1994, Lau, Kim and Shen 2002, Saha et al. 2014) which is related to the so-called spring barrier, a phenomenon of lower forecast skill for ENSO sea surface temperatures (SST) in the Equatorial Pacific for spring and summer conditions compared to fall and winter conditions (Barnston et al. 1994, Balmaseda, Davey and Anderson 1995, Wen, Xue and Kumar 2012, Beraki et al. 2014). Skill also varies geographically, depending on the time and location of occurrence of the dominant atmospheric signals, for example ENSO, North Atlantic Oscillation (NAO), or Pacific Decadal Oscillation (PDO) (Barnett et al. 1993, Davies, Rowell and Folland 1997, Muller, Appenzeller and Schar 2005, Jin et al. 2008, Polade et al. 2013). ENSO is widely assumed to have an important if not the most important impact on U.S. climate (e.g., Ropelewski and Halpert 1986). However, its impact is inhomogeneous across seasons and regions (Peng et al. 2012) which is reflected in the accuracy of forecasts for different regions. Kerr (2008) studied CPC's tercile forecasts and found that temperature forecasts were correct in more than 85 percent of cases "across much of the eastern [U.S.] out to more than eight months" (Kerr 2008 900) when an ENSO signal existed, compared to only 13 percent without an ENSO signal. Kerr (2008) also found that precipitation forecasts "along the southern tier states and up the West Coast" (Kerr 2008 900) were correct in 50 to 85 percent of cases "about half a year into the future" (Kerr 2008 900) during El Niño or La Niña phases, compared to 3 percent in years without a significant signal.

Skill maps and data accompanying NMME's ensemble forecasts for the U.S. often show low skill for much of the U.S. Great Plains, especially for precipitation (Kirtman et al. 2014). "High skill is evident in the central and eastern tropical Pacific Ocean, as well as portions of the tropical Atlantic and Indian Oceans and some isolated regions in the extra-tropics" (Kirtman et al. 2014 589). Wintertime extreme precipitation, which can be devastative to grain crops, is correlated to ENSO signals in the southeastern U.S., the Gulf Coast, central Rocky Mountains, and the Ohio-Mississippi River valleys and responds strongly enough to make it predictable (Gershunov and Barnett 1998). The ENSO signal itself (i.e. the occurrence of El Niño, La Niña, and neutral phases in the tropical Pacific) is quasi-periodical with dominant peaks occurring about every four years and minor peaks every two years (Latif et al. 1998). Jin et al. (2008) tested a multi-model ensemble and found an overall strong correlation (0.86) between predicted and observed ENSO state at six months lead, which was higher than any single model in their test. Jin et al. (2008) also found that strong ENSO events are better predicted than weak ENSO events, neutral phases are predicted worse than warm (El Niño) or cold (La Niña) phases,

and the skill of predictions that start in spring decreases faster than skill of predictions that start in fall.

3.2 Forecast effectiveness

Forecast value can also be expressed as a function of its value in management practices. Seasonal forecasts are useful for agricultural producers if they allow them to adjust crop- or livestock-related management decisions according to the forecast (Stern and Easterling 1999). Schneider and Garbrecht (2003a, 2003b, 2006) designed a method to assess the effectiveness of seasonal forecasts. In their context, effectiveness is a function of the deviation of the forecast from the long-term average. It represents the percentage of forecasts in a region that are considered above or below average and are forecasted in the correct direction (e.g., forecasted warmer than average when observations are warmer than average, too). Schneider and Garbrecht (2003a, 2003b, 2006) assumed that forecasts had more value to agricultural producers the more they deviated from climatology while being correct. They argued that decisions would have greater positive financial impacts (i.e., larger profits or smaller losses) the more that the forecasts deviated from the longterm average — an alternative basis for farm decision-making (Schneider and Garbrecht 2003a, Schneider and Garbrecht 2003b). From 1997 to 2005, effectiveness for CPC's 0.5month lead-time temperature outlooks was highest in the southwestern U.S., and high in the Pacific Northwest, parts of Texas, and the Florida Peninsula, as Fig. 4 (top) shows. For much of the remaining U.S., including most of the Great Plains, effectiveness was less than 20%. The 0.5-month precipitation outlooks show a similar pattern but an overall lower effectiveness, in particular across the agricultural regions of the Great Plains and the Midwest (Fig. 4, bottom). Garbrecht et al. (2010) conclude that such low occurrence

of forecasts with large deviations from climatology, especially in agricultural states like Oklahoma, discourage the development of decision-support tools for stocking rate selection in winter-wheat grazing operations.



Fig. 4: Effectiveness of seasonal temperature (top) and precipitation (bottom) forecasts for 0.5 months lead- time, for each of the 102 climate forecast divisions. Percentage values indicate what portion of forecasted values were above or below normal and in the same tercile category as the observed values. Source: Schneider and Garbrecht (2006)

4 Seasonal climate forecasts in the context of other farm decision factors

Agricultural decisions are in large steered by weather and climate conditions, but other factors play an important role, too, such as markets, costs, land constraints, or production goals (Klockow, McPherson and Sutter 2010). Jones et al. (2000) show that potential benefits from climate forecasts also depend on wealth and risk aversion of producers. Higher risk aversion is commonly associated with higher crop diversification, which in Jones et al. (2000) for the southeastern U.S., leads to reduced average annual farm income but also lower fluctuation in income. In other words, a crop mix that is adapted to the respective forecast slightly reduces average farm income but also substantially reduces potential financial losses while creating more planning security for producers. Forecast value increased with increasing risk aversion, particularly in cases of low initial wealth (Jones et al. 2000).

Meinke et al. (2003) suggest a systems approach by including crop simulation results, climate science, and systems analysis into discussions about risk mitigation and adaptation with stakeholders. Meinke et al. (2003) conducted a case study in India in which growers either adjusted plant density or crop type (i.e. planted sorghum or peanuts instead of cotton) in response to seasonal climate forecasts. The majority of farmers responded by adjusting in one way or the other and still made a harvest, but "ca. 20% of farmers, who took the risk and planted cotton [a crop with higher potential profits than sorghum or peanuts but less suited for the forecasted climate conditions] had to abandon their crops by August, losing all their input costs" (Meinke et al. 2003 4).

Harwood et al. (1999) found that changes in laws and regulations, decreases in crop or livestock production, and uncertainty regarding commodity prices are the biggest concerns of agricultural producers. While irrigation, crop insurance, enterprise diversification, or production contracting are ways to alleviate risk in crop production (Barham et al. 2011, Harwood et al. 1999), they are not always without drawbacks, like reduced income, additional costs, or loss of entrepreneurial freedom (see examples in Harwood et al. 1999 2-3). Other studies add to a broader picture of risks in farming and ranching, such as risks in human resources (e.g., lack of necessary farm labor) or marketing (variations in commodity prices and/or quantities that can be marketed), together with finance (maintaining income, avoiding bankruptcy), production (variations in yields due to weather, pests, and diseases), and legal actions (law suits over contractual agreements, government regulations about pollution and farm practices), which influence farmers' decision-making and suggest mitigation measures (Musser and Patrick 2002).

Meinke and Stone (2005) argue that probabilistic forecasts and proactive farming practices allow farmers to capitalize on beneficial seasonal conditions and buffer against detrimental ones. Choi et al. (2015) support this idea in an economic study from Spain on cotton, two vegetable crops, two grain crops, and animal calories. They concluded that revenue increases not only were positive for almost all crops using proactive measures, but were also remarkably higher (between 1.5 and 5 times) compared to conservative approaches, which also often had revenue losses (Choi et al. 2015).

Dual-purpose production of winter wheat and cattle is a common practice in the U.S. Southern Great Plains, as it provides a second source of income and spreads risks (Colorado State University 2010). It provides two management scenarios: growing winter wheat (1) solely as cattle feed or (2) for grain production with the option of cattle grazing during winter (in the early stages of crop growth) as nutritious hay supplement for cattle during this part of the year. Deciding which scenario to opt for is in large determined by market prices for cattle meat and wheat grain, but also by climate forecasts for the growing season. Differences in regional precipitation amounts or timing can translate into different crop yields for wheat and thus live weight gain for cattle. This means that depending on market prices for grains and livestock and the amount of growing season precipitation (November to March), either grain production or meat production offers the highest return-on-investment. For example, a study in northern Texas by Mauget et al. (2009) used different market prices and climate forecasts to show how market price conditions can dominate decision-making over using seasonal climate forecasts. If a respective climate forecast was not significantly different from normal such that best management practices would not change, then that seasonal forecast had no economic value (because it did not foster changes in practice). This concept of critical threshold was also used in studies mentioned earlier by Schneider and Garbrecht (2003a, 2003b, 2006) and Garbrecht et al. (2010) who found that for large parts of the Great Plains and the Midwest, seasonal climate forecasts for temperature only rarely deviated substantially from the long-term average and therefore provided only small additional value for decision-makers in this region (see Fig. 4). Mauget et al. (2009) also showed that depending on market conditions, best management practices did not always benefit from more localized seasonal climate forecasts. This meant that forecasts with higher spatial resolutions did not necessarily translate into higher forecast values for decision-makers.

5 Forecast demands and forecast communication

In line with the development and improvement of operational seasonal climate forecasts, climate services have also evolved over the past decades. The field of climate services focuses on two areas: (1) measuring, recording, and providing climate data, and (2) interpreting data to generate climate information (Changnon 2007). Climate information for U.S. agricultural producers can be communicated through agricultural advisors like the Cooperative Extension Service (CES), a network of state headquarters at land-grant universities and subordinate networks of county offices affiliated with the headquarters in their state (USDA 2015a). CES extension agents provide research-based advice for crop and livestock producers, such as weather- and climate-based information and agricultural practices (USDA 2015c). For agricultural producers, failing to adapt to a more variable future climate could result in lower yields (Meinke et al. 2003). To allow producers to benefit from favorable climate conditions and to reduce losses from unfavorable ones, it was proposed to develop tailor seasonal forecasts to specific user groups (e.g., Meinke et al. 2003, Lamb, Timmer and Lélé 2011), and numerous modeling studies and user surveys were conducted to determine current shortcomings and to explore user needs in more detail (Prokopy et al. 2013, Schneider and Wiener 2009, e.g., Fraisse et al. 2006, Hansen and Indeje 2004). Effective forecast communication is also discussed internationally. In a special issue of the WMO Bulletin on the "Global Framework for Climate Services", Tall (2013) proposes an interactive, five-step method to deliver tailored climate services to end users: 1) understanding the demand side, 2) bridging the gap between climate forecasters and sector expertise, 3) co-producing climate services to address end-user climate service needs, 4) communicating to reach 'the last mile', and 5) assessing and re-assessing.

5.1 Forecast demands from producers and agricultural advisors

Schneider and Wiener (2009 100A) list nine forecast requests based on a survey of agricultural producers and water managers:

- 1. forecasts of up to one year for long-term planning
- 2. forecasts for regions regarded as competitors
- 3. better warnings of anomalous events such as snow storms in spring or flash floods
- 4. more clear explanations and documentation of the accuracy and reliability of data and forecasts
- 5. information provided in a "now versus last year versus normal" format
- 6. what weather patterns and storm tracks commonly recur in the region

7. the need for simple procedures to "calibrate" large-scale forecasts and warnings to local areas

- 8. better information to improve decision-making related to wildfires
- 9. observations and forecasts on soil moisture and relative humidity

Most producers use seasonal climate forecasts to improve planting schedules, for irrigation and nutrition management, and to select crop type and crop variety (Cabrera et al. 2006, Templeton et al. 2014). They request seasonal climate forecasts of both direct meteorological variables (e.g., air temperature and precipitation) and derived information from these and other variables (e.g., humidity, growing degree-days, soil moisture, or evaporative loss) (Schneider and Wiener 2009, Frisvold and Murugesan 2012). Tab. 1 links some of these variables with agricultural decisions.

Tab. 1: Agricultural decisions connected to different types of meteorological variables. Source: Frisvold and Murugesan (2012)

Type of weather data	Agricultural decision	
	Planting, harvesting, defoliation, crop modeling,	
Temperature	disease risk, shelter animals, pest control, sheep	
	shearing	
	Planting, harvesting, fertilizer applications, cultivation,	
Precipitation	spraying, irrigation, disease risk, livestock and poultry	
	protection	
	Planting, harvesting, fertilizing, transplants, spraying,	
Soil Moisture	irrigation, monitoring of growing conditions,	
	measuring plant stress	
Soil Temperature	Planting, pest overwintering conditions, transplanting,	
	fertilizing	
	Pest overwintering conditions, Protect crops from	
Frost	damage, animal sheltering, irrigation (to avert crop	
	damage)	
Degree Days	Planting, irrigation, pest control	
Belative Humidity	Harvesting, pollination, spraying, drying conditions,	
The active Fluctuates	crop stress potential	
	Defoliation, harvesting, freeze potential/protection,	
Wind Speed	animal sheltering, shelter, pest control, pruning,	
	spraying or dusting, pollination, dust drift, pesticide	
	drift	
Wind Direction	Freeze potential/protection, cold or warm air	
	advection over crop areas, pesticide drift, dust drift	

Different crops, such as winter wheat, cotton, sugarcane, or corn, are planted and harvested in different months, and require different precipitation patterns, different management practices, and different decisions at certain times of the year (Steiner et al. 2004). Tailored climate forecasts should therefore be crop-specific and depend on time of year and decision lead-time (e.g., days or months) (Meinke et al. 2003, Steiner et al. 2004, Meinke and Stone 2005, Haigh et al. 2015). Crop-specific calendars (Tab. 2, Tab. 3, Fig. 5) reflect this need, inform about critical crop stages, and highlight when weather and climate information can be particularly helpful in making crop-related decisions. For example, Fig. 5 show that soil temperature information is particularly important for corn producers during winter when fertilizer is applied that could volatilize if temperatures are too high (Takle et al. 2014). Tab. 3 shows how production and marketing decisions for corn differ depending on lead-time and season. Likewise, Tab. 4 shows how decision leadtimes differentiate between short-term (operational) and long-term (strategic) crop-related decisions, such as sowing depth, planting time frame, crop type and variety, or decisions on equipment purchase (Hudson 1972). Finally, Mavi and Tupper (2004) illustrate the benefits of key management decisions for livestock and various grain crops, demonstrating why seasonal climate forecast are important, and what management decisions can help mitigate unfavorable conditions.

Tab. 5 lists key management decisions for livestock and various grain crops, demonstrating why seasonal climate forecasts are important and what management decisions can help mitigate unfavorable conditions.

A growing number of publications call for comparisons to previous years (e.g., Schneider and Wiener 2009) and the need for seamless short-term weather to seasonal climate forecasts (NOAA 2011, Tall 2013). Information on forecast reliability and accuracy are also requested by producers (Schneider and Wiener 2009, Takle et al. 2014) and thus are needed to communicate predictability to the users. Unclear accuracy levels

Month	Crop and Stage (summer crop)	Crop and Stage (winter crop)
January	Sugarcane: harvesting	Grains: dormant
February	Sugarcane: harvesting	Grains: dormant
March	Sugarcane: harvesting Cotton: planting	Grains: vegetative
April	Corn, small grains, cotton: planting	Grains: vegetative
May	Corn, small grains, cotton, sorghum: planting	Grains: heading
June	Small grains: heading Soybeans: planting Corn, cotton, sorghum: vegetative	Grains: maturing to harvesting
July	Small grains: filling to maturing Corn: silking Soybeans: flowering Sorghum: heading Cotton: blooming	Grains: harvesting
August	Small grains: harvesting Corn, soybeans, sorghum, cotton: filling	-
September	Small grains: harvesting Corn, soybeans, sorghum, cotton: maturing	Grains: planting
October	Corn, soybeans, cotton: harvesting	Grains: vegetative
November	Corn, soybeans, cotton, sugarcane: harvesting	Grains: hardening
December	Cotton, sugarcane: harvesting	Grains: dormant

Tab. 2: Summer and winter crop stage calendar for the US. Source: USDA (2015b)

and perceived inaccuracy are top barriers for producers, scholars, extension agents, and consultants to use seasonal climate forecasts (George et al. 2007). To overcome these barriers, survey respondents suggested more localized and more accurate forecasts with higher degrees of certainty, information about past accuracy, and provision of skill scores (George et al. 2007).



Fig. 5: Climate-based decision cycle for corn. The outer calendar identifies the time of year management decisions are made. The inner calendar depicts the soil or crop impact, and the label on the arrow identifies the weather or soil conditions relevant to the impact Length of the arrow gives the lead-time of climate forecasts that links the specific agricultural decisions to soil or crop impacts. Source: Takle et al. (2014)

	March – May	June, July	August – October	November – February
Production				
Decision	Field work, planting	Pest/disease/weed/ irrigation management	Harvesting, grain drying	Seed variety, fertilizer, crop insurance
Real-time Monitoring	Soil moisture, wet days	Wind, humidity, soil moisture	Soil moisture	
1-3 days	Soil moisture, soil temperature	Humidity, wind, rainfall	Wet days	
1-3 weeks		Spring soil moisture, summer humidity, summer drought, flood	Fall freeze	Rain (fall), Snow (winter)
1-6 month		Summer drought, extreme daytime heat, fall freeze, fall wet days		Spring drought, flood, soil moisture, temperature
6-12 months				Summer drought/flood
Marketing				
Decision	Purchase/sell	Purchase/sell corn	Purchase/sell	Purchase/sell
Decision	corn in storage	in storage	corn in storage	corn in storage
Real-time monitoring	Spring freeze	Global drought, flood, extreme daytime/ nighttime hear, hail/wind damage	Fall wet days	
1-3 days				
1-3 weeks	Late spring freeze	Soil Moisture, drought, flood, extreme daytime/ nighttime heat		
1-6 months	Summer drought, flood,	Summer drought, extreme daytime/ nighttime heat, fall freeze, fall wet days		Spring drought, flood, soil moisture, temperature
6-12 months				Summer drought, flood, extreme daytime/ nighttime heat

Tab. 3: Corn growers' needs for forecast information at specific times of the crop year. Source: (2011)

Tab. 4: Farm management decisions related to different forecasts lead-times. Source: Hudson $\left(1972\right)$

Time frame	Agricultural aspects to consider
	which depth to plant the seeds
	whether to sow a crop or not
	deciding whether or not to irrigate
	deciding whether or not to harvest crop
5-day forecast	• time spray programs to ensure maximum efficiency against pests and
	diseases affected by weather
	frost protection of crops
	increase efficiency of herbicides
	decide on supplementary pollination due to poor weather
	extend the time of seeding to cover successional sowings of crops
	like peas, to ensure a steady flow to the processing plant
	harvest crops for short-term storage in cases when conditions might
	interfere with lifting (e.g. frozen soil) or prevent harvesting (e.g. soil
	soft and saturated with water)
	• put perishable goods in short-term storage, or on the market, based
5-week forecast	on estimates of demand and supply
	• decide on water use, based on the probability of non-moisture-stress
	days
	decide on dates for sowing crops sensitive to frost, to maximize the
	chance of achieving a large leaf-area index by the time light
	conditions are optimal
	deciding on straying, depending on whether or not diseases and
	pests will affect crops
	decide on growing marginal crops depending on upcoming seasonal
	temperatures, e.g. tomatoes, which are highly profitable in a hot
	summer but fail entirely in a cool summer
	managing scarce water resources
5-month	 employing and scheduling labor for handling crops, determining
forecast	cropping schedules
	match fertilizer application to expected yields
	planning timely measures against diseases and pests likely to affect
	crops based on seasonal forecast
	choose crop varieties based on seasonal forecast
	estimate acreage for a required tonnage of crop
	lay out long-term path of crop choices and alternatives based on
	expected weather and climate shifts
5 year forecast	• plan long-term investments for special equipment, e.g. for harvesting,
	irrigation, frost protection, short-term storage, grain drying
	(for policy-makers) determine national policy of support and
	development of marginal crop areas

Tab. 5: Farm management decisions related to seasonal climate forecasts. * indicates decision that are also influenced by forecasts of shorter time scales, such as extended weather forecasts or short-term weather forecasts. Source: Excerpt from Mavi and Tupper (2004)

Key Decision	Why seasonal climate	Strategies to reduce	
Rey Decision	information is important	losses/increase profits	
Management			
Investment in new machinery	Purchase/hire of high-cost machinery requires good weather for maximum income to ensure easy repayment.	Make large purchases in seasons when the outlook is normal or better than normal.	
Seasonal planning	Warmer weather conditions may cause crops to mature early. Excessively wet season requires planning for control of weeds, insect pests, and diseases.	Book labor and contractors earlier to harvest crops.	
Cropping	1		
Crop variety to plant	Most crop species have a number of varieties available that vary in their length of growing season or resistance to heat, cold, frost, water logging, or disease.	Choose a crop variety that best suits the seasonal conditions. Plant varieties that mature before the possibility of late frost. Plant a long-season variety of rainfall is likely to be evenly spread and a short-duration variety of probability is of less rainfall.	
Fertilizing	Fertilizing with nitrogen can increase crop yields potential but only if there is sufficient rainfall.	Fertilize at the optimum rate only if the outlook for the season is favorable.	
Disease control	Many crop diseases are affected by weather. As an example, yellow spot in wheat can become prevalent in wet years, causing reduced production.	Be prepared for disease control of the outlook is for a wet season. Monitor the crop and undertake a spray application when the first symptoms of disease become apparent.	
Weed control*	Wetter years or wetter than average seasons may cause an increase in the number of crop weeds.	Spray earlier to ensure weeds don't get too large, and if using ground spraying, spray when damage to soil by machinery is least.	

Tab. 5: Continued

Sugarcane			
Replant or retain old ratoon	New plantings culminate in poor stands and stunted growth in dry seasons	New planting should take place only in a favorable season. Maintain old ratoon if conditions are unfavorable.	
Trash blanket	Cover ground with trash in dry weather to preserve soil moisture.	Do not burn trash in dry years; harvest green.	
Viticulture			
Harvesting	Warm temperatures enhance growth and harvesting is easier	Plan to harvest earlier if seasonal outlook is of warm weather.	
Water / Irrigation			
Water allocations	Weather will determine if storage or water source is replenished	Crop smaller areas when outlook is for dryer conditions and water allocation is low. Adopt water-saving practices.	
Stock water*	Hot, dry weather increases stock water intake and increases evaporation from the stored water	If the seasonal outlook is for lesser rain, use water sparingly and budget water allocation between animals and paddocks.	
Grazing / Pasture			
Optimum stocking rates*	Climate determines the type and amount of grass and herbage growth.	If seasonal forecast is favorable, stocking rates can remain at current levels.	
The number of stock to carry during the dry season	Weather determines how much stock feed will be available.	Lower stock numbers before dry conditions set in to avoid cost of feeding or sale of stock at low market prices.	
Burning pasture for weed control*	Weather affects the effectiveness and safety of using fire as a tool. In the longer term, burning before a dry period may mean a shortfall in feed supplies.	Burn grass only on days with low fire danger; burn only small areas if the outlook is poor, so that there will be extra feed for dry periods.	
Fire breaks*	Weather can affect the severity of the fire season leading up to fire occurrences.	Maintain fire breaks early in the season and increase preparedness on potentially dangerous days.	
Feeding and supplements	Dry periods result in little or no plant growth	Budget to feed or supplement stock; buy and stockpile feed.	

Tab. 5: Continued

Pasture improvement*	Pasture improvement is a costly program, and the aim is to maximize establishment of pasture. Ideal climatic conditions are required for pasture improvement.	Undertake pasture improvement if seasonal forecast is favorable.
Haymaking		
Marketing	Hay prices are usually low in good seasons and high in poor seasons.	Stockpile hay if the outlook is for a dry season and sell in the dry seasons at better prices.
Sheep and Wool		
When to shear*	Choose a time of year to shear when newly shorn sheep are not subject to extreme weather changes.	Shear when rainfall in less likely or when major temperature changes do not occur. Increase area under cover for sheep.
Supplementary feeding	Lack of rain may necessitate early feeding of costly supplements to maintain growth and minimize production losses.	Decrease stock numbers; buy feed supplements earlier at lower prices. Feed early to minimize losses.
Treatment for fly control*	Warm humid weather increases incidence of sheep becoming struck/infested with flies.	Treat sheep with chemical before problems occur, or monitor sheep carefully in susceptible periods.
Footrot*	Wet conditions favor spread of footnote in sheep.	Plan to have sheep in paddocks where they are less susceptible to prolonged wet conditions.
Parasite control	Wet conditions allow an increase in the level of internal parasites.	Pasture sheep in paddocks with less possibility of wet soil; drench sheep to decrease worm numbers coming into a wet season.
Cattle		
Restocking	After drought, producers often buy stock to take advantage of extra paddock feed.	Restock only if seasonal outlook is favorable. A break in the season may not last long, necessitating early sale or feeding of stock, causing losses.

Tab. 5: Continued

Weaning	Calves may need to be weaned off their mothers earlier if there is a dry period and then sold or fed.	Weaning calves early in dry weather stops stress on cows and allows them to go into calf for the following season.
Parasite control	In wet conditions internal worms are more likely to increase in numbers.	Treat stock early to avoid buildup of parasites, or pasture in areas where parasites are not such a problem.

5.2 Communication Challenges

Mismatched terminology, unrealistic expectations, and disordered integration of information into the decision process create a communication barrier between scientist and stakeholder (Lemos and Rood 2010, Briley, Brown and Kalafatis 2015). In order to satisfy end users, scientists need to do research with the end user in mind (Hartmann 2002). Education of researchers about decision-making in agriculture could increase the applicability of forecast products for agricultural producers (Takle et al. 2014).

To explain the value that a user perspective can add to providing forecast advice, Hubbard (2007 2) gives a hypothetical dialogue between a crop producer (client) and a public service employee (expert), for example an agricultural extension agent:

Client: I need temperature data, do you have a station near Westbend?

Expert: What is the question you are trying to answer?

C.: I want to know the length of the freeze free season so I can decide whether to buy the new corn hybrid offered by my seed dealer.

E.: We can provide the average length of the growing season but, we also can provide information on the variability. So we can provide answers to the question 'How many years out of 10 would the growing season exceed 175 days?'. Would you be interested in this analysis?

C.: Yes, I knew that each growing season was different but, I didn't know how to include that in addressing my problem.

E.: We can also show that years with the same number of growing season days often have a difference in heat available, or GDDs [Growing Degree Days], to move the crop to maturity. We can assess the GDDs by year and give you information to answer how many years out of 10 does the accumulated GDD exceed a threshold, e.g. 2800.

C.: That would be great.

This dialogue illustrates the importance of a common mindset among expert and client (compare Schneider and Wiener 2009). Hubbard (2007) emphasizes that the expert in this example not only answers the initial question but points out additional information to answer the client's request. The expert's language resembles that of the client when communicating advice.

To improve the value of forecasts, Takle et al. (2014) list questions that forecast developers should know the answers to:

• What meteorological variables are needed to improve the climate-related decisions? Is there linkage of this information to remote, slow time-varying forcing such as ENSO, the Atlantic multi-decadal oscillation (AMO), the Pacific decadal oscillation (PDO), and soil moisture?

• At what points in the annual or inter-annual decision cycle are these variables needed?

• How can past information best be collected and archived for effective data mining?

- Who makes the decisions and when; what is the lead-time?
- Are combinations of meteorological conditions important in certain cases, such as consecutive days of extreme rainfall and high winds followed by drought, which could lead to toxins in crops, e.g., aflatoxins in corn, that can be harmful if consumed by livestock or humans (compare Vincelli, Parker and McNeill 1995).
- What ancillary biological or soil information is needed such as crop development stage, plant physiology, soil fertility, terrain slope, weeds, insects, and diseases to allow evaluation of both biotic and abiotic impact on the crop?
- What else could help translate meteorological data into decision aids, e.g., crop growth/yield model, soil compaction model, soil erosion model, and calculation of days per week suitable for field work?
- What is the best way to convey uncertainty metrics: graphs, tables, PDF, or terciles of skill?

Intermediary organizations that act between scientists and stakeholders can help bridge the various gaps in language and mindset. Examples for these intermediaries are the Department of the Interior (DOI) Climate Science Center network (DOI 2009), the U.S. Department of Agriculture Regional Climate Hubs (Allen and Stephens 2016), and the NOAA Regional Integrated Sciences and Assessments (RISA) teams (Lemos and Morehouse 2005, Miles et al. 2006), such as the Climate Assessment for the Southwest (CLIMAS), the Southern Climate Impacts Planning Program (SCIPP) or the Great Lakes Integrated Sciences and Assessment (GLISA) program (Briley et al. 2015).

Recently, much effort is put into web-based applications and services that presume computer and internet access among the agricultural community (e.g., Breuer et al. 2008, Fraisse et al. 2006, Pasteris, Puterbaugh and Motha 2004). A biennial survey by the U.S. Department of Agriculture (USDA), based on about 28,000 responses from farmers and represents all sizes and types of farms within the contiguous U.S. (USDA 2015b), found that internet access is common among farmers (70% in 2015). However, it is unclear from the survey how the internet is accessed. In 2015, only 43% of all U.S. farmers use computers in their farm business (USDA 2015b). While smartphone use is not assessed in the survey, web-based applications that require computers could miss a large portion of the U.S. agricultural community.

6 End-to-End Concept

Traditionally, climate forecast products have been developed in two ways: (1) by using existing forecast information for practical uses (top-down approach), such as CPC's categorical seasonal forecast, or (2) by taking user demands and finding niches for climate



Fig. 6: Schema of the end-to-end concept. Actionable outcomes require interaction and feedback from societal, biological, and physical elements. Source: Goddard et al. (2001)

forecasts to assist in decision-making, such as the NMME seasonal multimodel ensemble forecast for soil moisture, runoff, as well as skill information (bottom-up approach) (Goddard et al. 2001). Both approaches are restricted in their potential because they emphasize some components while neglecting others. An end-to-end concept encompasses the entire work process, from forecast production, tailoring and communicating the forecast, to forecast application by end users, while also considering social behavior, institutional constraints, or sector system models (Agrawala, Broad and Guston 2001, Goddard et al. 2001, Power, Plummer and Alford 2007). Fig. 6 illustrates the end-to-end concept. End-to-end forecast development is being spurred by existing relationships between climate science and various other fields, such as agriculture, food security, disaster management, disease incidence, and disease risk (Coelho and Costa 2010). Development of end-to-end products is achieved by interacting and collaborating with user groups, conducting workshops and surveys in order to understand their needs, learning their vocabulary, and framing the limits of their understanding when it comes to weather and climate (Bales et al. 2004). Coelho and Costa (2010) present a "simplified framework" for an end-to-end forecast system, consisting of three key elements: (1) the underlying climate forecast information ("climate science"), (2) the impacts of climate on human and natural systems ("systems science"), and (3) decision-making that is "performed on the basis of forecast information jointly produced by climate and systems sciences" (Coelho and Costa 2010 318). Shafiee-Jood et al. (2014) apply an end-to-end approach in a case study of corn farmers in Illinois, studying the value of seasonal climate forecasts during an extreme drought in the summer of 2012. Garbrecht and Schneider (2007) review the top-down and participatory end-to-end approach for agriculturefocused climate forecasts and discuss a hybrid method of forecast development using a participatory approach and forecast dissemination following a top-down approach.

Once an end-to-end forecast system is established, a recurring dialog between forecast users and forecast producers needs to take place to continually improve climate information and its use (Bales et al. 2004, Roncoli et al. 2009). Within the U.S., RISA (Lemos and Morehouse 2005), administered by the NOAA Climate Program Office, facilitates this discussion through regional initiatives that provide actionable science for a range of decision types, from agriculture to natural hazards mitigation. RISA initiatives with an agricultural focus include CLIMAS (Bales et al. 2004), GLISA (Briley et al. 2015), and the Southeast Climate Consortium (SECC; Fraisse et al. 2006). Similarly, an international example for (trans-disciplinary) integration of scientific knowledge and user input to assist decision-making is the work of the International Research Institute for Climate and Society (Mason et al. 1999, Barnston et al. 2003, Lyon et al. 2014).

7. Conclusion

Seasonal climate forecasting has come a long way over the last centuries and improved substantially in recent decades. Thanks to a better understanding of atmospheric processes, advances in computing, and improved prediction models, seasonal forecasts of temperature and precipitation are now a standard forecast product available in the U.S. and many other countries around the world. A new challenge of making these forecasts more valuable to specific users, like agricultural producers, is now being approached by integrating social science and climate science. Researchers found an increasing appreciation of seasonal climate forecasts by producers despite still essential shortcomings, and they are beginning to understand farmers' decision-making processes and decisiontiming. As a result, forecast developers are able to better transform basic forecast data into tailored forecast products for specific sets of decisions and degrees of comprehension, in order to improve the value of seasonal forecasts. The current state of the science are endto-end concepts of continuous development and feedback loops, integrating both the development of prediction models and tailoring forecasts according to user needs.

8. Acknowledgements

The authors thank two anonymous reviewers for their critiques and comments, which enhanced this paper. This work was funded by the Office of the Dean of the College of Atmospheric and Geographic Sciences and by the Office of the Vice President for Research at The University of Oklahoma, and by the Oklahoma EPSCoR project through the Oklahoma State Regents for Higher Education and the National Science Foundation under NSF grant no. OIA-1301789.

9. Conflict of Interest Statement

The authors have no conflicts of interest to declare for this study.

Chapter 3 - Assessing Decision Timing and Seasonal Climate Forecast Needs of Winter Wheat Producers in the South-Central United States

Submitted as:

Klemm, Toni & Renee A. McPherson: Assessing Decision Timing and Seasonal Climate Forecast Needs of Winter Wheat Producers in the South-Central U.S. Journal of Applied Meteorology and Climatology.

Abstract

Agricultural decision making that adapts to climate variability is essential to global food security. Crop production can be severely impacted by drought, flood, and heat, as seen in recent years in parts of the United States. Seasonal climate forecasts can help producers reduce crop losses, but many nationwide, publicly available seasonal forecasts currently lack relevance for agricultural producers, in part, because they do not reflect their decision needs.

This study examines the seasonal forecast needs of winter wheat producers in the Southern Great Plains to understand what climate information is most useful and what lead-times are most relevant for decision making. An online survey of 119 agricultural advisors in Oklahoma, Kansas, Texas, and Colorado was conducted that gave insights into producers' preferences for forecast elements, what weather and climate extremes have the most impact on decision making, and the decision timing of major farm practices. Winter wheat growers were not only interested in directly modeled variables, such as total monthly rainfall, but also derived elements, such as consecutive number of dry days. Extension agents perceive that winter wheat producers needed seasonal climate forecasts to cover one to three months in advance — the planning lead-time for major farm practices, like planting or applying fertilizer. We created a forecast calendar and monthly rankings for forecast elements that can guide forecasters and advisors as they develop decision tools for winter wheat producers and can serve as a template for other time-sensitive decision tools developed for stakeholder communities.

1. Introduction

Droughts, floods, heat waves, extreme rainfall, or other unseasonable weather and climate conditions can have considerable impact on agricultural productivity and farm revenues, and they are costly for the taxpayer. From 2011 to 2014, when large portions of the United States (U.S.), in particular the Great Plains and the Midwest, were hit by severe drought and flood, federal crop insurance paid an average of \$12.4 billion — three times the annual average from 2001 to 2010 (USDA 2016a, USDA 2016b) — to compensate farmers for losses in crop yields. Seasonal climate forecasts can help crop producers make better educated decisions (Carberry et al. 2000, Meinke et al. 2003) that reduce or prevent revenue losses due to unseasonably warm, cold, dry, or wet conditions or even allow producers to capitalize on these conditions and increase yields and revenues (Carberry et al. 2000, Jones et al. 2000, Meinke and Stone 2005).

The U.S. Climate Prediction Center (CPC) has been producing seasonal climate forecasts since the 1940s, constantly improving forecast skill and increasing forecast leadtime (van den Dool 1994, O'Lenic et al. 2008). A better understanding of atmospheric and oceanic processes, like El Niño (e.g., Ropelewski and Halpert 1986), and improvements in computing power have also advanced seasonal forecasting, not only in the U.S. but worldwide. Currently, 12 countries and multinational organizations around the world issue global seasonal climate forecasts for environmental variables that include air temperature, sea-surface temperature, and precipitation, following standards set by the World Meteorological Organization (WMO 2015). For example, CPC issues monthly seasonal climate forecast ensembles based on eight individual models, with up to seven months lead-time for average, maximum, and minimum air temperature, precipitation, and soil moisture as well as other variables (Kirtman et al. 2014).

For decades, seasonal climate forecasts have been of interest to the agricultural industry for decision support (Changnon et al. 1988, Sonka et al. 1992, Changnon 2004). Farmers and ranchers use these forecasts to improve key decisions, such as irrigation scheduling, planting, harvesting, fertilizing, or selecting crop type and crop variety (Cabrera et al. 2006, Frisvold and Murugesan 2012, Templeton et al. 2014). Although usage of seasonal climate forecasts by agricultural producers has increased, complaints regarding lack of skill and lack of lead-time were common (Changnon 2004). A comprehensive review of the development of seasonal climate forecasting for agricultural producers, including examples of knowledge co-production and communication challenges between forecasters and forecast users, can be found in Klemm and McPherson (2017). Schneider and Wiener (2009) concluded that there is a lack of mutual understanding between the forecast and user community, leading to a lack of relevance of produced forecasts for decision making of farmers and ranchers. Recent studies pointed out that crop-specific seasonal forecasts, for example for corn farmers, could improve farm decision making related to weather- and climate-related risks (Takle et al. 2014). Our study tackles these challenges for a single crop and region so that the detailed timing of production decisions can be documented as they relate to weather and climate impacts.

2. Methods

The goal of this study builds upon forecast limitations and needs previously identified; it can be summarized by this research question: "How can seasonal climate forecasts be tailored to serve the needs of winter wheat growers in the south-central United States?" To answer this question, an online survey was sent to agricultural extension agents to study decision-making patterns in winter wheat production and the specific forecast needs of these producers. This study also explored the specific role of agricultural advisors in the decision-making process of winter wheat growers. Ultimately, the intent was to create a foundational understanding from which to develop decision-support tools based on the needs, timing, and professional network of the user rather than the capabilities of the provider.



Fig. 7: The study region. Extension agents from counties with diagonal lines participated in the survey. Yellow/green shaded counties show the acreage of planted winter wheat in 2016. Acreage data source: USDA National Agricultural Statistics Service

2.1 Why winter wheat?

This study focused on one particular crop type (as opposed to all grain crops) because decision making, and especially the timing of decisions, is different from one crop to another. Winter wheat is the dominant crop in the Southern Great Plains of the U.S.. In the study region, the states of Colorado, Kansas, Oklahoma, and Texas (Fig. 7), winter wheat is grown on 21.1 million acres (in 2016), more than twice the acreage of the second largest crop, corn (Han et al. 2012). Wheat itself, used for both food and livestock feed, is the world's largest crop by harvested acreage (FAO 2014). In the U.S., wheat is the third-largest crop after corn and soybeans, and winter wheat contributes about 70 percent to the total wheat harvest in the U.S. (USDA 2012). Because of its overall contribution to the national wheat harvest and its role in the study region, winter wheat was chosen as focus crop for this study.

2.2 Survey population and distribution

Data for this study were collected via an online survey using the Qualtrics survey platform. The survey was distributed to agricultural advisors, specifically cooperative extension agents associated with land grant universities in the study domain (Fig. 7). Approximately 360 cooperative extension agents were contacted and 119 unique responses (ca. 33 percent response rate, average for online surveys, according to Nulty (2008)) were received. 10 of these responses stated no winter wheat was grown in their jurisdiction and were discounted. 109 were eventually used in the analysis. Extension agents are agricultural advisors with academic backgrounds who work on a county level and advise producers on best practices.

Extension agents were chosen as survey participants rather than winter wheat producers themselves because the former represented a more homogenous group regarding educational background and access to email and internet, reducing representation bias and the risk of using unfamiliar terminology. Each agent also works with a large number of farmers in their jurisdiction; thus our results represent the aggregate views of the producer community.

Online surveys have been shown to be more efficient than phone or in-person interviews for quantitative or binary (yes/no) questions (Babbie 2014), such as those used in this survey. Survey distribution is inexpensive, easy, and fast via email, and survey responses are available in electronic format, eliminating transcription errors and postprocessing time. Survey sponsors, people known in the extension community (i.e., extension district directors and state climatologists), helped distribute the survey. They received email address lists and text templates, including the survey URL and a one-page summary about the survey. Some survey sponsors did not use the email list but instead sent the invitation/reminder via their own mailing lists. The authors were copied on all emails to record send dates and times and to get confirmation about the sending. After initial survey invitations were distributed, two to four reminder emails were sent out, with intervals of three to four weeks in between each reminder.

Procedures by Dillman et al. (2014) were adopted to increase survey responses and to enhance the quality of the responses. "Survey sponsors" (Dillman et al. 2014) sent initial survey invitations and later reminders to extension agents in their jurisdictions. After pretesting the survey with three extension agents, incorporating revisions, and receiving Institutional Review Board approval, initial survey invitations were sent between 19 January 2016 (Oklahoma) and 2 March 2016 (Colorado). The survey closed on 6 May 2016.

2.3 Survey design

The survey time was estimated to take 10-15 minutes to complete. Median response time was 12 minutes. 130 individual responses were received. After eliminating double and triple responses (only the first responses were kept) and empty forms, and responses from 10 agents who stated no winter wheat was grown in their jurisdiction, 109 responses were used in the analysis (see Tab. 6). Eighteen respondents did not include state or county information. Those responses were used in all analyses except to examine regional differences in responses.

State	Number of responses
Colorado	3
Kansas	23
Oklahoma	37
Texas	28
No location information	18
Total	109

Tab. 6: Number of survey responses by state and number of responses without location information.

Location data based on IP addresses for these responses were available; however, it was found these data to match the actual locations poorly in the cases that included state and county information, and therefore IP addresses was not used. For nine responses, agents entered more than one county as their area of responsibility. For these cases, their jurisdiction was treated as a single, large region as opposed to separate responses for every county they stated. Thus, the regional analysis sample was comprised of responses from 99 individuals who provided location information (state and county, or region). To geolocate these responses, the center coordinate of the respective county or region was calculated using QGIS, an open-source geo-analysis software, using the *centroid* function.
Using this center coordinate, responses were sorted by latitude and longitude to examine regional differences. All further statistical analysis was done in Microsoft Excel[®] for Mac.

The survey consisted of 16 questions (Appendix 1) that were developed to answer the research question — How can seasonal climate forecasts be tailored to serve the needs of winter wheat growers in the south-central United States? — and to provide insights into the farming communities in the study region. The first seven questions collected general information about the extension agents and the winter wheat producers they serve. The next two questions asked about their familiarity with producers' information needs and the level of influence of weather and climate information on the advice that extension agents give to their winter wheat producer clientele. Both questions helped determine the importance of monthly and seasonal forecast timescales for agents' advice overall. Question 10 described the agents' professional communication network. Question 11 measured agents' levels of agreement with various statements related to seasonal forecasting and climate variability and extremes, with responses to this question describing the needs of their producers. Questions 12 and 14 asked about agents' knowledge of *when* farmers make decisions on specific farm practices and their perception of *what* forecast elements could assist which decisions. These responses were used to create a tailored forecast calendar (section 3.4). Question 13 asked agents about weather threats affecting producers' long-term decisions, information that can be used to tailor forecasts to inform specifically about extreme conditions like drought, late frost, or heat, which can negatively affect crop growth. Lastly, the final two questions requested voluntary contact information for follow-up interviews (which were not conducted due to time constraints) and additional comments. See the appendix for the full survey.

Prior to designing the survey, a short list was created of forecast elements desired by agricultural producers and feasible to be provided by current forecast models. The authors consulted with agricultural educators from the Oklahoma Mesonet (McPherson et al. 2007) and extension employees prior to survey design. In addition, relevant literature related to agricultural surveys (Mavi and Tupper 2004, Cabrera, Letson and Podestá 2007, Schneider and Wiener 2009, Prokopy et al. 2013, Takle et al. 2014) and seasonal forecast model development (Jia et al. 2015, Jia et al. 2016, Kirtman et al. 2014, Vecchi et al. 2014) was examined in order to ask questions that explored user needs while also recognizing current seasonal forecast capabilities.

3. Results

3.1 General survey statistics

On average, participants in the extension agent survey had been working in their current state for about 15 years (the range was zero to 42 years). Forty-two percent of respondents indicated that 1–49 percent of producers in their county grew predominantly winter wheat; 48 percent of agents said that percentage was 50 or higher. Nine percent responded that no one grew predominantly winter wheat; these participants did not take part in the remaining survey. One percent did not know. The majority of advisors (87 percent) answered that producers grew predominantly unirrigated winter wheat, and about half (46 percent) answered that 50 percent or more of their producers grow dual-purpose winter wheat, which, unlike grain-only winter wheat, is also used as feed supplement for cattle in winter.

Extension agents indicated that current conditions and 1-7 day forecasts were the most relevant lead-times for their advice to farmers, with 56 percent and 50 percent of



Fig. 8: Survey responses from extension agents about the level of relevance of different weather or climate forecast information for their advice to crop producers. responses, respectively (Fig. 8). The longer the lead-time, the less influential the forecasts were, agents responded. Monthly and seasonal lead-times had "large influence" for 24 percent of agents. It is unclear, however, if this result is because of a greater importance of shorter lead-times or because of the lack of confidence in forecasts with longer lead-times. Weather data for the past 12 months and historical weather trends fared second lowest in priority, behind all but annual to longer-term forecasts. In their efforts to network, extension agents are most often in contact with farmers (90 percent on a daily or weekly basis) and least often with state climatologists and the State Department of Agriculture, with whom some agents have never interacted at all. Between these extremes and at comparable levels are seed producers, farm chemical dealers, other advisors, and farm organizations such as the Farm Bureau.

Agents were asked about their level of agreement with two statements about climate variability. For the first statement, 70 percent of extension agents agreed or strongly agreed that "In the last five years, I've seen more variability (e.g., more extremes) in the climate across my county." Only 4 percent disagreed or strongly disagreed with this statement. Fifty-five percent of agents agreed or strongly agreed that "Climate variability

hurts my growers more than it benefits them" while 2.5 percent disagree or strongly disagree. Interestingly though, in the former question, the rate of agreement increased with an increase in the years of work experience, even though the question only asked about the past five years. A similar trend of increasing agreement with increasing work experience was detected in the latter question. Thirty-three percent disagreed or strongly disagreed with the statement that "Current seasonal forecasts are insufficient for winter wheat producers," while 26 percent agreed or strongly agreed.

3.2 Decision timing

Planning patterns for most decisions were unimodal or bimodal (Fig. 9) meaning that agents indicated specific decisions were made only once or twice per year (Tab. 7 and Fig. 10). For example, August 30 was noted as the peak time for planning to plant winter wheat, May 28 for harvest planning, and July 24 for purchasing a specific crop variety.



Which months do winter wheat growers make decisions on the following specific decisions? (n = 109)

Fig. 9: This matrix shows when winter wheat growers plan for certain agricultural practices. The calendar months are listed on the x-axis, labeled at the bottom; practices are listed on the y-axis, labeled on the right. The size of the bubble represents how often a given practice was selected for that given month by extension agents, with larger bubbles representing more responses. Overall, the graph depicts the seasonality of important, climate-related decisions in winter wheat production.

Bimodal decision peaks during spring and fall referred to practices that are conducted twice a year; for example, planning peaks for fertilizing occur on February 12 and September 5. To determine the average planning dates for bimodal planning decisions, the respective distribution was split at its minima into two sub-distributions, then calculated the peak time and time range for each of the two resulting unimodal sub-distributions. The sub-distributions did not always have the same lengths. Following this process, peak times were March 23 and October 4 for disease control planning, March 4 and September 17 for weed control, and April 1 and September 26 for irrigation scheduling. As Fig. 9 shows, planning for water resource management showed no particular peak but appeared relevant all year round. For this reason, it was excluded From Fig. 10 and from some of the further analyses.

Regional differences in the timing of decision peaks or decision time spans were calculated by splitting all responses into roughly equal-sized subgroups, four by latitude

Tab. 7: Date of peak planning times and average decision time span, calculated by creating the average of the last month extension agents selected for every decision and subtracting it from the average first month agents selected for every decision.

Decision	Decision peak and average decision time span			
Planting timing	August 30, 1.35 months			
Harvest timing	May 28, 1.1 months			
Purchase crop variety	July 24, 1.45 months			
Fertilizer application	Spring: February 12, 0.71 months	Fall: September 5, 1.08 month		
Disease control	Spring: March 23, 1.27 months	Fall: October 4, 1.09 month		
Weed control	Spring: March 4, 0.93 months	Fall: September 17, 1.58 months		
Growing cover crops	June 11, 1.95 months			
Managing water resources	No peak, year round			
Seasonal employment	June 20, 3.7 months			
Irrigation scheduling	Spring: April 1, 1.68 months Fall: September 26, 1.52 months			
Grazing scheduling	December 15, 3.55 months			

and four by longitude. A t-test was conducted to test for statistical significance at the 95% level of differences in decision timing. In most cases differences in decision timing were insignificant and/or inconsistent. However, in four cases, decision peak time or decision time span shifted statistically significantly. For harvest timing, peak decision time shifted by 28 days from south to north, from May 11 to June 8. Peak decision timing for fall fertilizing shifted by 35 days from south to north, from September 29 to August 25. Peak decision timing for spring disease control shifted by 37 days from south to north, from March 5 to April 11. And for seasonal employment, the decision time span increased by 93 days from west to east across the study area, from 66 days to 159 days.

As expected, these planning-time patterns aligned with the timing and seasonality of the actual decisions. Unexpectedly, however, survey results suggested a relatively short lead-time for climate information ahead of the decisions. For example, planting for winter wheat takes place between early September and early October (Colorado State University 2010, Kansas State University 1997). With the planning-time peak for planting in late August and an average time span of 1.35 months, the required average lead-time for climate forecast products was about 0.5 months. Similarly, harvest planning peaks on May 28, on average about 1.5 months before harvest time. Taking into account the time range in responses, the preferred lead-time for forecasts to inform harvest planning is zero to 2.5 months.



Fig. 10: This diagram, based on the responses shown in Fig. 9 and Tab. 7, represents the timing of when survey participants expected winter wheat producers to make key decisions (e.g., application of weed control) for their farm. Calendar months are represented as pie slices, labeled near the center of the diagram. Light gray ticks on the circle denote one fifth of each month. Average decision periods and decision peaks for farm practices are highlighted by yellow and blue circle segments; words within the circle and yellow or blue bars mark peak times. The average winter wheat growing season is included in green for comparison. It starts in September (Planting) and ends in July (Harvest). The dotted green circle segments indicate average time of planting and harvesting. The figure suggests that average planning horizon for decisions (and thereby lead-time for seasonal climate forecasts) is zero to 2.5 months. Water resource management is not included because it did not show a particular seasonality.

3.3 Weather and climate threats

Survey results suggested that drought was the number one weather or climate threat overall, and it was connected to more decisions than any other listed threat (Fig. 11). (Threats were ranked by counting how many "yes" responses they received.) Drought was followed by extreme rainfall, heat, wind/storm, frost, and hail, in that order. Some decisions were more affected by threats than others. For example, planning for harvest timing was most sensitive to unseasonal weather conditions overall, followed closely by planting timing. Least affected were planning for growing cover crops and seasonal employment. Fig. 11 also shows that drought and heat were considered greater threats during planting timing as compared to harvest timing, likely because seeds need moist soil



Fig. 11: This matrix shows what management practice decisions are impacted by which weather and climate threats. The practices are listed on the x-axis, labeled at the bottom; threats are listed on the y-axis, labeled on the right. The size of the bubble represents how often a given practice was selected for that given month by extension agents, with larger bubbles representing more responses. The ranking of threats is based on the total number of times they were associated with a practice. It shows that the most impactful threat is drought, followed by extreme rainfall. The matrix also highlights that some threats impact some practices more than others, e.g., hail impacts harvest time more than any other listed practice.

to germinate. Storms and hail, on the other hand, were greater problems during harvest planning, when the matured wheat plant can be easily damaged by either. Extreme rainfall was a major concern for both planting and harvesting because it can make fields inaccessible for the necessary planting and harvest machinery. "Does not apply" was least selected, suggesting that all listed threats were, in some way, relevant for these decisions. At the same time, the extension agents did not use the text boxes for entering additional threats.

3.4 Forecast preferences

Lastly, extension agents were asked what forecast information can help improve what specific decision or decisions. The intent was to diagnose (1) what forecast information is most and least important to wheat farmers, and (2) what seasonal forecast elements should be provided by seasonal climate forecasts. Overall, all forecast elements related to precipitation were ranked higher than elements related to temperature (Fig. 12). Average precipitation ranked highest, followed by consecutive days without rainfall, deviation from average precipitation, and chances for extreme rainfall. In fifth place, average air temperature was the highest-ranked temperature-related forecast information. Perhaps surprisingly, growing degree days, used to estimate plant growth and maturation based on air temperature in horticulture and agriculture (Bonhomme 2000), only ranked 9th out of 11.

Finally, to ensure that products were tailored for south-central U.S. wheat producers, requests for specific forecast elements (e.g., average monthly precipitation) were ranked by month (Tab. 8). For each calendar month, this ranking was calculated by multiplying the number of survey responses per month for each management decision (i.e., the



Which forecast information can assist which decision(s)? (n = 109)

Fig. 12: This matrix shows which forecast elements help producers plan for which practices. The practices are listed on the x-axis, labeled at the bottom; forecast elements are listed on the y-axis, labeled on the right. The size of the bubble represents how often a given practice was selected for that given month by extension agents, with larger bubbles representing more responses. The ranking of forecast elements is based on the total number of times they were associated with a practice. Average precipitation is the most helpful forecast element overall, and the top four forecast elements all relate to precipitation. Ranks 5 to 11 relate to temperature.

underlying data for Fig. 9) by the associated number of responses per forecast element for each management decision (i.e., the underlying data for Fig. 12). By doing so, the relative importance of each forecast element was calculated for each calendar month. These calculations were summed by calendar month and forecast element, and the forecastelement totals were ranked for each calendar month, as shown in Tab. 8. Average precipitation ranked first and consecutive days without precipitation ranked second throughout the year. The ranking of the remaining forecast elements varied from month to month.

						œ	anking	-					
Forecast Element	Avg.	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
Average precipitation	-	۲	1	-	1	1	۲	1	-	1	1	1	1
Consecutive days without precipitation	2	2	2	2	2	2	2	2	2	2	2	2	2
Deviation from average precipitation	3	3	4	4	4	4	4	3	3	3	3	3	3
Chances for extreme precipitation	4	4	3	3	3	3	3	4	2	4	4	4	4
Average temperature	5	5	5	5	5	9	9	5	4	5	5	5	5
Consecutive days above 100°F	9	6	9	9	6	5	5	6	9	9	9	6	6
Average maximum temperature	7	8	7	7	7	7	7	6	6	8	8	6	6
Growing degree days	8	7	8	8	8	8	8	8	8	6	6	7	8
Deviation from average temperature	6	10	6	6	6	6	6	7	7	7	7	8	10
Average minimum temperature	10	11	11	10	10	10	10	10	10	10	10	11	11
Consecutive days below 32°F	11	o	10	11	11	11	11	11	1	11	11	10	7

Tab. 8: Ranking for forecast elements based on monthly decision-timing and forecast preferences for each decision.

4. Discussion and Conclusion

This paper summarized results from an online survey of extension agents in the Southern Great Plains about decision timing and seasonal forecast needs in winter wheat production in their jurisdiction. It was found that most management decisions addressed by the survey occurred with a distinct seasonality once or twice per year, and they peaked from one to three months before the respective practice was conducted. In agreement with existing literature (Mavi and Tupper 2004), it was found that forecast elements based on precipitation were more relevant to producers than those based on temperature. Somewhat surprisingly, forecasts for growing degree days, specifically developed for farming and horticulture, did not rank highly at all. Decision timing varied across the study region, but apart from four cases, it occurred without statistically significant spatial trend.

Despite the seasonality of most management decisions, the two most requested forecast elements — average precipitation and consecutive days without precipitation remained as the highest priorities throughout the year while others changed ranks from month to month. Comparing the northern (eastern) vs. southern (western) part of the study region, some of the rankings changed, indicating that forecast providers should keep forecasts regionally relevant. That said, though, the authors suggest that operational seasonal forecasts be designed for existing administrative regions, such as the National Weather Service forecast areas, to better fit into existing distribution networks and to minimize additional operational expenses for issuing these forecasts.

Overall, results suggested that winter wheat producers plan for the surveyed subset of management decisions (e.g., planting, harvesting) only one to 2.5 months before

operationalizing those plans (Kansas State University 1997, Colorado State University 2010). These lead-times are well within the capabilities of current models in seasonal forecasts, such as the North American Multimodel Ensemble, with up to 7 months leadtime (Kirtman et al. 2014), or the Geophysical Fluid Dynamic Laboratory's Forecastoriented Low Ocean Resolution (FLOR) model, with up to 12 months lead-time (Jia et al. 2015, Vecchi et al. 2014). In many cases, shorter forecast lead-times have higher forecast skill than longer lead-times (Kirtman et al. 2014), particularly for precipitationrelated forecast elements, which ranked highest in priority in our study. An exception to this rule is the so-called "spring barrier," which limits the skill of seasonal summer forecasts regardless of lead-time because of higher uncertainty in forecasting the Equatorial Pacific conditions that control summer climate variations in many parts of the world, including the U.S. (Barnston et al. 1994, van den Dool 1994, Balmaseda et al. 1995, Lau et al. 2002, Wen et al. 2012, Beraki et al. 2014, Saha et al. 2014). The desire for shorter lead-times also suggests that tailored seasonal forecasts may have sufficient skill at time scales relevant for decision making, helping to address producers' complaints of the past (Changnon et al. 1988, Sonka et al. 1992, Changnon 2004, Schneider and Wiener 2009).

Surveying extension agents has advantages, as explained in section 2, but also some limitations. Extension agents, as all human respondents, might have been biased in their responses and based their answers on recent memories or on interactions with peers, for example (Nadeau and Niemi 1995). Extension personnel were also one step removed from the decision makers (i.e., winter wheat producers) themselves. For example, extension agents can say little about the rationale for individual farm management decisions and the factors that contributed to it, such as the timing of a decision, why they preferred one forecast element over another, or how important weather and climate forecasts were relative to other decision factors, such as markets, costs, or production goals (Klockow et al. 2010). In addition, survey respondents only represented growers who actually interact with their local extension officials, which might have created bias and potentially left out a considerable part of the winter wheat community. Thus, the analysis and insights were limited by the knowledge that our participants had about their clients. The survey was also unable to say whether the preference for a zero to 2.5 month decision lead-time was caused by the limited forecast skill of current seasonal climate forecasts or because of other management issues. In other words, would the decision lead-time have been longer if producers thought that more skillful forecasts were available? As a result, we recommend ongoing communication between the climate forecast community, the agricultural extension community, and the producer community so that forecast improvements can be incorporated effectively into decision tools, and decision tools can be adjusted based on decision making and forecast availability.

Regardless of how good the tailored seasonal climate forecasts are or can be, growers may choose not to apply them because of conflicts with other decision factors, including other climatic factors. For example, winter wheat growers can delay planting in case of dry soil when rain is forecasted soon; however, planting cannot be delayed too much or otherwise the plant might not mature enough to survive the cold winter. Likewise, if planting occurs too early (because of suitable conditions earlier than normal), increased growth before winter dormancy can deplete soil moisture too much which jeopardizes growth in spring and eventually a good harvest (Kansas State University 1997). Finally, the findings of this study apply strictly to winter wheat farming in the Southern Great Plains. Different crops, such as cotton or corn, or different winter wheat regions around the world can have very different decision calendars and therefore require different tailored forecasts.

Despite these limitations, these results can provide climate forecasters with information that can help address criticism of seasonal forecasts from the agricultural community mentioned at the beginning of this paper. The results have fundamental value in communicating user needs to forecasters and forecast model developers. The results provide insights into the timing of major long-term decisions in winter wheat farming and suggest ways in which forecasters can adjust or create seasonal forecasts to serve the needs of these producers and assist them in making proactive management decisions to reduce crop losses as a result of unseasonal weather and climate conditions.

5. Acknowledgements

We thank the USDA Cooperative Extension Services of Texas, Oklahoma, Kansas, and Colorado, and the state climatologist of Texas for participating in this survey and for their invaluable help pretesting and distributing it. Thanks also to Drs. Mike Langston and Jean Steiner for connecting us to the extension community, and to Al Sutherland and Gary McManus for their feedback on early versions of the survey. We also thank Drs. Derek Rosendahl, Duncan Wilson, Elinor Martin, Esther Mullens, Jason Slemons, Joe Ripberger, and Travis Gliedt, as well as Mr. Benjamin Ignac for their help in our analysis and visualization. This work was funded by the Oklahoma EPSCoR project through the Oklahoma State Regents for Higher Education and the National Science Foundation under NSF grant no. OIA-1301789, with additional support from the Office of the Vice President for Research of the University of Oklahoma (OU) through the South Central Climate Science Center and the Dean of the College of Atmospheric and Geographic Sciences through the OU Department of Geography and Environmental Sustainability.

Chapter 4 - Assessing Model Accuracy of Seasonal Climate Forecasts for Winter Wheat Producers in the South-Central United States

To be submitted as:

Klemm, Toni & Renee A. McPherson: Assessing Model Accuracy of Seasonal Climate Forecasts for Winter Wheat Producers in the South-Central United States.

Abstract

Seasonal climate forecasts have been used by the agricultural community for decades in decision planning and have been critiqued, for example, for their lack of skill or lack of relevant information. This study analyzed whether a high-resolution, single-model seasonal climate forecast, tailored to the needs of winter wheat producers in the southcentral United States, is more accurate than a 5-year seasonal persistence forecast.

Average monthly temperature, surprisingly, was almost never forecasted more accurately by the model, and the model error was particularly high in summer at long lead-times. Model forecasts for average monthly precipitation showed strong seasonality and were more accurate than persistence during winter across in large parts of Texas for all lead-times (zero through 11). On the other hand, the number of dry days and the number of days with extreme precipitation per month were forecasted more accurately by the model compared to persistence for summer across large parts of the study area and less accurately than persistence in winter. Overall, both extreme precipitation amounts and the number of dry days per month were vastly underestimated by model and persistence forecasts.

1. Introduction

For decades, the agricultural community has been using long-term weather and climate forecasts to inform decision making and to warn for unseasonal conditions (Klemm and McPherson 2017), for example, for decision planning on irrigation, planting, harvesting, or trading of livestock (Frisvold and Murugesan 2012). However, despite using these forecasts, producers raised critiques about numerous shortcomings, such as low accuracy, not enough forecast lead-time (Sonka et al. 1992), or a general lack of understanding in user needs (Schneider and Wiener 2009); thus, they often preferred to use historical records, year-to-date information, or other observation-based information for decision making (Changnon et al. 1988, Haigh et al. 2015). Moreover, Klemm and McPherson (2017) pointed out that publicly available seasonal forecast information by the National Weather Service's Climate Prediction Center (CPC) is not tailored to specific user needs, such as those of agricultural producers.

As a result, numerous studies addressed existing shortcomings by improving seasonal forecast models (O'Lenic et al. 2008, Delworth et al. 2012, Kirtman et al. 2014, Saha et al. 2014, Jia et al. 2015). Other studies explored forecast needs through collaboration with users and decision makers, for example in the southeastern United States (U.S.) (Breuer et al. 2006, Cabrera et al. 2006), the Midwest (Takle et al. 2014, Haigh et al. 2015), or the south-central U.S. (Klemm and McPherson in review), and how to improve communication pathways to decision makers (Dilley 2000, Hansen 2002, Lemos, Kirchhoff and Ramprasad 2012, Taylor, Dessai and de Bruin 2015, Allen and Stephens 2016). Qualitative studies described the complexity of decision processes in cattle production (Wilmer and Fernández-Giménez 2015, Wilmer et al. 2016). In addition,

boundary organizations were created in the U.S. to build relationships with climate information users and to provide capacity and expertise for this type of research, such as the National Oceanic and Atmospheric Administration (NOAA) Regional Integrated Sciences and Assessment (RISA) teams (Miles et al. 2006, Lemos and Morehouse 2005), the Climate Science Center network of the U.S. Department of the Interior (DOI 2009), or the U.S. Department of Agriculture Climate Hubs network (Allen and Stephens 2016).

The study presented here is a continuation of a survey conducted by Klemm and McPherson (in review) about decision timing and forecast needs of winter wheat producers in the south-central United States (U.S., see chapter 3). We used their results (see Tab. 2) to set priorities in an analysis of the accuracy of a numerical climate model compared to a persistence forecast of the same spatial resolution (50 km by 50 km). Our intention was to explore the capability of a forecast model to provide tailored information of both averages and extremes requested by producers, according to the survey. Existing literature (see above) described general skepticism of the accuracy and quality of seasonal climate forecasts and the preference of users for historical and observed data. This study was intended to compare accuracy of a model forecast and a persistence forecast, which is derived from observational data, to assess if a model forecast could be a more accurate alternative to a persistence forecast for forecast elements of interest to winter wheat producers in our study area.

2. Study area, data, and methods

In this section, we address our dataset choices and explain our data processing procedures, including how and why we chose certain criteria and thresholds.

2.1 Study area climatology

Generally, monthly total average precipitation in the study area, a four-state region in the south-central U.S. comprised of Texas, Oklahoma, Kansas, and Colorado, ranges from 34 mm (January) to 85 mm (May); average monthly temperature ranges from 4 °C (January) to 26 °C (July) (Fig. 13). From October to March, precipitation is highest in eastern Texas and the Colorado Rocky Mountains and lowest in western Texas and eastern Colorado. From April to September, the precipitation peaks over the eastern half of the study area, especially eastern Oklahoma and Kansas, and decreases westwards. Temperature follows a south-north gradient all year, with maxima being observed in southern Texas and minima being observed in the Colorado Rocky Mountains.

2.2 Observational and model datasets

For our analysis we used a total of six temperature (monthly) and precipitation (daily and monthly) datasets: three gridded, observational datasets and three model datasets, all of which had the same temporal and spatial resolutions and same time period (Tab. 9). Using the National Oceanic and Atmospheric Administration (NOAA) Earth System

Dataset	Variables	Spatial Resolution	Temporal Resolution	Time period	Lead time
GHCN CAMS	Air temperature	0.5° x 0.5°	monthly	1980-2014	-
CPC Global Unified Precipitation	Total Precipitation	0.5° x 0.5°	daily	1980-2014	-
NOAA PREC/L	Total Precipitation	0.5° x 0.5°	monthly	1980-2011	-
GFDL FLOR B01	Air temperature	0.5° x 0.625°	monthly	1980-2014	0-11 months
GFDL FLOR B01	Total Precipitation	0.5° x 0.625°	daily	1980-2014	0-11 months
GFDL FLOR B01	Total Precipitation	0.5° x 0.625°	monthly	1980-2011	0-11 months

Tab. 9: Description of the observational and model datasets used in the study.



Fig. 13: Observed averaged for (A) monthly total precipitation and (B) monthly average temperature across the study area. Solid lines represent averages, dashed lines represent minima and maxima.

Research Laboratory Physical Science Division dataset catalogue, we selected the Global Historical Climatology Network CAMS dataset for average monthly air temperature (Fan and van den Dool 2008), the CPC Global Unified Precipitation dataset for daily total precipitation (Xie et al. 2007, Chen et al. 2008), and NOAA's Precipitation Reconstruction over Land dataset for monthly precipitation (Chen et al. 2002).

Using the National Center for Atmospheric Research Earth System Grid data catalogue (<u>www.earthsystemgrid.org</u>), we obtained corresponding model temperature (monthly) and precipitation (daily and monthly) hindcast data from the Geophysical Fluid Dynamics Laboratory (GFDL) Forecast-oriented Low Ocean Resolution (FLOR) model, variation B01. This non-flux-adjusted climate model output 12 historical simulations, all of which use identical model physics but were initiated with different initial conditions (Vecchi et al. 2014, Jia et al. 2015, Jia 2017). Seasonal forecasts are issued every month for the coming 11 months, starting with the month of issuance (lead zero) to 11 months out (lead 11). For example, the seasonal forecast issued in January 2010 covered January 2010 to December 2010. Issuing forecasts in the case of our study ended in December 2010. This, however, means that for every month in 2011 the number of forecasts decreased from January to December.

A seasonal persistence forecast (hereafter simply called persistence forecast) was used to compare against the monthly model forecasts. It was calculated as the unweighted average of the preceding five years for the respective month of interest. For example, the persistence forecast for December 1985 was the average of observations in December 1980, December 1981, December 1982, December 1983, and December 1984. For the analysis of daily data, the same was done in a daily fashion. As a result, the comparison period started in 1985 and ended one year prior to the end of the data record (2010 for monthly precipitation and 2013 for all other forecast elements).

2.3 Survey data input

The forecast elements prioritized for this study were based on survey data collected by Klemm and McPherson (in review) from 109 agricultural advisors in Texas, Oklahoma, Kansas, and Colorado. The study focused on decision timing and seasonal forecast preferences of winter wheat producers in the south-central United States (U.S.), where winter wheat is the dominant crop type. The researchers found that decisions on practices like planting, harvesting, or fertilizing had a strong seasonality and were planned for only in certain times of the year, specifically about one to three months before they were carried out. Klemm and McPherson calculated a ranking of 11 forecast elements and found that producers' decisions most often required forecasts of average total precipitation, consecutive days without precipitation, the deviation from average precipitation, chances for extreme precipitation, and average temperature (Tab. 8). We analyzed model performance on four of the top five forecast elements for each month: average precipitation, consecutive days without precipitation, chances for extreme precipitation, and average temperature. Because of our method of analysis, we decided to leave out deviation from average precipitation.

Our goal was to evaluate model accuracy for forecast elements most relevant to winter wheat producers in the south-central U.S. while also accounting for basic forecast abilities of a seasonal forecast model. Therefore, we decided to interpret "consecutive days without precipitation" and "chances for extreme precipitation" as "number of days per month with no precipitation" and "number of days per month with extreme precipitation" because (1) it is highly unlikely that on a daily resolution with lead-times of several months, extreme precipitation occurrence or no precipitation occurrence will be forecasted correctly, and (2) knowing months in advance exactly when extreme precipitation will occur with unknown skill levels does not provide producers with actionable information.

In the past, various definitions have been applied to define extreme precipitation and no precipitation. Groisman et al. (2005) considered the top 0.3% of daily rainfall events in a study of Northern Hemisphere changes in intense precipitation. Zhai et al. (2005) used the 95th percentile of days with rainfall of all weather stations for a trend analysis of extreme precipitation frequency in China between 1951 and 2000. Higgins et al. (2011) took into account the top 50 daily events when studying the frequency of extreme precipitation events between 1950 and 2009. Villarini et al. (2013) analyzed extreme precipitation over the U.S. Great Plains using a peak-over-threshold approach with the 95th percentile of non-zero precipitation values for each of the 447 weather stations in their study, accounting for regional differences in the absolute values of the threshold. Schoof and Robeson (2016) used a percentile classification developed by the Expert Team on Climate Change Detection and Indices (ETCCDI) under the umbrella of the World Meteorological Organization Commission for Climatology/World Climate Research Programme project on Climate Variability and Predictability; these values are commonly referred to as ETCCDI indices (Schoof and Robeson 2016).

In our analysis, we used ETCCDI indices on a grid-cell basis, rather than a fixed precipitation threshold (e.g., 50 mm/day) or a predetermined number of extreme events (e.g., the 100 highest precipitation days) as "they are easy to interpret and are directly related to impacts in agriculture and other sectors" (Schoof and Robeson 2016, p. 29). Defining threshold by grid cell also acknowledged the strong precipitation gradient across the study region, from the Gulf of Mexico in the southeast to the Rocky Mountains in the northwest (see PRISM Climate Group). In accordance with ETCCDI, we chose the 95th (99th) percentile of all days (January 1st 1985 to December 31st 2013) with precipitation equal or greater than 1 mm as the threshold for very (extremely) wet days, and days with less than 1 mm of daily precipitation as dry days (Tab. 10).

2.4 Data preparation and processing

All data processing was done using the NCAR Command Language (NCL), netCDF Operators (NCO), and bash shell scripting, with the occasional use of Microsoft Excel for

Classification	Definition
very wet days	95th percentile of days with \geq 1 mm precipitation
extremely wet days	99th percentile of days with \geq 1 mm precipitation
dry days	days with < 1 mm precipitation

Tab. 10: Definitions or precipitation thresholds used in the study.

testing and verifying procedures. Model and observational data were obtained for a global domain for the time periods of interest. Observations were subset using a shapefile of the four-state study region (Texas, Oklahoma, Kansas, and Colorado) created in QGIS with data from Global Administrative Areas (GADM, <u>www.gadm.org</u>), a web portal for geographic shapefile datasets. Model data was first subset to a latitude-longitude rectangle around the study region with about 1.5° buffer around the four states and then regridded using bilinear interpolation in NCL ("linint2") from its original 0.5° x 0.625° grid to a 0.5° $\ge 0.5^{\circ}$ grid that matched the observational datasets in location and spatial resolution. After bias correcting the model data (see next section), we calculated the differences between each forecast dataset and its observational counterpart for each forecast month, lead-time, and model run. We also calculated the difference between each persistence forecast and its associated observations for a given month. For example, the persistence forecast for December 1985 is the average of observations in December 1980, December 1981, December 1982, December 1983, and December 1984. Finally, we compared the absolute differences between model and observations (absolute model error) and persistence and observations (absolute persistence error) to assess whether forecast or persistence had smaller errors. This approach is similar to calculating a skill score.

We used a factor-based bias correction for precipitation and a difference-based bias correction for temperature, as described by Maraun et al. (2010) and Crochemore et al. (2016), respectively, to correct systematic model biases. We calculated the bias at lead

zero for each calendar month and removed this monthly bias from all lead times (zero to 11) for the respective month. Biases were computed for each grid cell and forecast month (January to December) by comparing the lead zero forecast for each month and year to their observational counterparts and then averaging the differences by forecast month. For precipitation, we *divided* each lead zero forecast value for every grid cell by its corresponding observational measurement; a quotient greater (smaller) than one meant over-prediction (under-prediction). These quotients were averaged by forecast month and represented the relative bias at lead zero for a particular forecast month, which we subsequently used for bias correction. For temperature, we *subtracted* monthly, individual observations from their lead-zero model counterpart before averaging these error by forecast month to obtain the lead-zero bias for every month, subsequently used for bias correction.

Graphs in Fig. 14 show minimum, maximum, and average values for precipitation and temperature bias for every forecast month. Precipitation bias showed two distinct patterns. From November to April, over-prediction of precipitation prevailed throughout most of the study region, with maxima in the western part of the domain and minima in the southeast that became an under-prediction during November and December. From May to September, the model ensemble under-predicted precipitation across large parts of the region, in particular the eastern half.



Fig. 14: Biases at lead zero for (A) monthly total precipitation and (B) monthly average temperature by forecast month. Note that the y axis in (A) is logarithmic. Errors above one (zero) indicate overestimation of the model for precipitation (temperature); errors below one (zero) indicate underestimation of the model for precipitation (temperature). Solid lines represent monthly averages, dashed lines represent minima and maxima across the study region.

Temperature biases showed three patterns. From December to April, over- and under-prediction occurred across roughly equal portions of the study region, with overprediction covering Kansas and Oklahoma and parts of Colorado and Texas, and underprediction occupying the remaining study region. From May to October, temperatures across most of the study region were over-predicted, except for parts of southern Texas and Colorado. In November, temperatures across most of the study region were underpredicted, with the exception of northwestern Kansas, eastern Colorado, and isolated spots in Texas and Oklahoma.

After determining monthly biases, we used linear scaling (Maraun et al. 2010, Crochemore et al. 2016) with precipitation forecasts, dividing each forecast datasets by their corresponding biases to create a bias-corrected precipitation. To correct temperature forecasts, we used a delta method (Maraun et al. 2010) and subtracted the bias from forecasts for corresponding months.

We also applied the factor-based method to correct *daily* precipitation. However, the patterns of these were very different from those of the monthly precipitation. The bias calculation determined that most of the model data at lead zero vastly overestimated precipitation, and correction therefore resulted largely in reducing the modeled values. This reduction, however, changed the model forecast values so much that for some grid cells all data were below 1 mm. Consequently, these data became unusable for creating thresholds for the 95th and 99th percentile (see below), because 1 mm was the threshold above which the 95th and 99th percentile were calculated. As a result, daily precipitation data were not bias-corrected.

3. Results

In this section we present and describe the results of the comparison of absolute model forecast error and absolute persistence forecast error to answer our research question: "Can existing seasonal forecast models provide meteorological variables as requested by winter wheat farmers with better accuracy than a persistence forecast?" We present results first for monthly total precipitation, then average air temperature, the number of days per month with extreme precipitation, and finally the number of days per month less than 1 mm of precipitation (dry days).

3.1 Total monthly precipitation

Observed average monthly precipitation between 1985 and 2011 had a bimodal distribution in our study area, with a seasonal pattern of local maxima in spring and fall and local minima in summer and winter. This result is in agreement with other datasets, e.g., from the National Centers for Environmental Information. The pattern was also



Fig. 15: Average total precipitation per calendar month for 1985 to 2011, for observations (black double line), persistence forecast (red dashed line), and leads zero to 11 of the biascorrected model forecast (lead zero to five: blue solid lines, lead six to 11: dotted grey lines). Note: The differences between observations and forecasts do not reflect the absolute errors described in the text. The persistence forecast in this case is essentially a climatological average.

present in both persistence and model forecast (Fig. 15). Monthly averages of the persistence forecast were roughly equal to observations, due to the way they were calculated.

The bias-corrected model forecast generally underestimated precipitation at shorter lead-times in all months and over-estimated it at longer lead-times during summer. We also found that the uncorrected data over-estimated monthly summer precipitation, albeit not as much as the bias-corrected data. It should be noted that the differences between average observations and average forecasts in Fig. 15 are different from the absolute errors and absolute error differences discussed in the following sections. While differences between model forecasts and observations in Fig. 15 reflected averages of the actual model error that included magnitude and direction of the error, averages of the absolute errors only included the magnitude, but not the direction of the error.

The following analysis was done using model data that was not bias-corrected. Comparing absolute model and absolute persistence errors, shown in Fig. 16, two distinct patterns emerged: in summer and early fall (June to September), the absolute persistence error was smaller nearly throughout the study area (green shaded areas in Fig. 16). In late fall and winter (November to February), considerable parts of the study area showed smaller model error (blue shaded areas in Fig. 16: Maps showing the differences between absolute model and absolute persistence forecast errors for every target month, averaged for all lead-times. Green (blue) shaded areas indicate smaller persistence (model) error.). Spring and fall were transition periods. Fig. 16 shows that during summer, averaged over all lead-times, the absolute error of the persistence forecast was smaller across the entire study area, especially in June and July over the western parts of Texas, Oklahoma, and Kansas. In winter, particularly in November and December, the model error was smaller in parts of Texas, Oklahoma and Kansas.

Differences in the spatial distribution and magnitudes of error differences between absolute persistence error and absolute model error, averaged by target month, were similar across different lead-times (Fig. 17), suggesting a seasonal character of the model





Мау



-2 -1.5 -1 -0.5 0 0.5 1 1.5 2



-2 -1.5 -1 -0.5 0 0.5 1 1.5 2





June Range: -0.04 - 2.02 m











April

August

Range: -0.12 - 2.46 mm/day Mean: 0.63 mm/d

Range: -0.07 - 0.7 mm/day Mean: 0.21 mm/da



July



-2 -1.5 -1 -0.5 0 0.5 1 1.5 2



-1 -0.5 0 0.5 1 1.5 2



-1 -0.5 0 0.5 1 1.5

Fig. 16: Maps showing the differences between absolute model and absolute persistence forecast errors for every target month, averaged for all lead-times. Green (blue) shaded areas indicate smaller persistence (model) error.

86

forecast. This result suggests that the absolute model error is more dependent on the target month than on the forecast lead-time. An exception in our analysis was the lead-zero forecasts for summer, when the error differences across the study area were considerably smaller compared to other leads of the same month (Fig. 17, bottom). This result could be due to two reasons: (1) large parts of the study area were generally under-predicted by the model forecast at lead zero but over-predicted at lead-times longer than four months, or (2) our bias correction method (see section 2.4), which generally reduced the model error at lead zero, but generally increased model errors at longer lead-times, especially in summer. One possible reason for this abnormal error at short lead-times is initialization shock (Jia 2017), which can occur in non-flux-adjusted models like GFDL's FLOR model, whereby energy fluxes of atmospheric and ocean components of a coupled forecast model are not in sync at initiation (Diro 2015, Mulholland et al. 2015). Consequently, using lead zero as a basis for bias correction of all lead-times would increase the error of the remaining lead-times. Fig. 18 shows that error differences were similar for leads zero through 11 for forecasts for November to May, with average differences fluctuating by less than 0.1 mm/day between lead zero and lead 11 in these months. Forecasts for June to October had larger average changes (0.16 to 0.46 mm/day) and a larger increase in error difference with increasing lead-time.

Time series from 1985 to 2011 showed year-to-year variations in error differences but no clear pattern across months or multi-year trend (Fig. 19). Some years stood out by having very small or even negative error differences at lead zero and lead 11 (for example February 1988, April 1998, or October 1999, 2001, and 2003) while pre- or succeeding years had considerably larger differences. Fig. 19 also illustrates how dissimilar the absolute error differences were are at lead zero and lead 11 for summer and early fall as compared to forecasts for winter and early spring.





-2 -1.5 -1 -0.5 0 0.5 1 1.5 2

April, lead 0 Range: 0.06 - 0.92 mmiday Mean: 0.29 mmiday

-2 -1.5 -1 -0.5 0 0.5 1 1.5 2

July, lead 0



-2 -1.5 -1 -0.5 0 0.5 1 1.5 2





·2 ·1.5 ·1 ·0.5 0 0.5 1 1.5 2

April, lead 4 Rage: 0.13 - 0.76 mm/day Mean: 0.19 mm/day

·2 ·1.5 ·1 ·0.5 0 0.5 1 1.5 2

July, lead 4



-2 -1.5 -1 -0.5 0 0.5 1 1.5 2







April, lead 8 Range: 0.07 - 0.63 mm/day Mean: 0.2 mm/day

-2 -1.5 -1 -0.5 0 0.5 1 1.5 2

July, lead 8



-2 -1.5 -1 -0.5 0 0.5 1 1.5 2





-2 -1.5 -1 -0.5 0 0.5 1 1.5 2

April, lead 11 Range: -0.09 - 0.68 mmiday Mean: 0.19 mmidag



July, lead 11



-2 -1.5 -1 -0.5 0 0.5 1 1.5 2

Fig. 17: Maps showing the differences between absolute model and absolute persistence forecast errors for forecasts for November, April, and July at lead zero, four, eight, and 11. Green (blue) shaded areas indicate smaller persistence (model) error.



Fig. 18: Time series showing the differences of absolute model and absolute persistence forecast errors from January to December from lead zero to 11. The thick line represents the average, thin lines the minima and maxima across the study area for a respective lead-time. Values above (below) zero indicate smaller absolute persistence (model) error.



Fig. 19: Precipitation time series from 1985 to 2011 comparing averages of absolute model and absolute persistence forecast errors for every target month. Graphs show the difference between absolute model and absolute persistence error at the shortest (thin line) and longest lead (thick line). Values above (below) zero indicate smaller model (persistence) error.
3.2 Average monthly air temperature

Observed average monthly temperature had a unimodal pattern in our study area between 1985 and 2013, with a maximum in summer and minimum in winter. Like average total precipitation, this climatological pattern agreed with other datasets, e.g., from the National Centers for Environmental Information. The pattern was replicated by both persistence and bias-corrected model forecasts (Fig. 20). Similar to monthly precipitation (section 3.1 and Fig. 15), monthly averages of the persistence forecast for temperature were also similar to monthly averages of observations. The bias-corrected model forecasts generally underestimated temperature at shorter lead-times in all months and over-estimated it at longer lead-times during summer. It should be noted that the error differences shown in Fig. 20 between average observations and average forecasts are not the same than the differences in absolute error discussed here.



Fig. 20: Average temperature per calendar month for 1985 to 2013, for observations (black double line), persistence forecast (red dashed line), and leads zero to 11 of the biascorrected model forecast (lead zero to five: blue solid lines, lead six to 11: dotted grey lines). Note: The differences between observations and forecasts do not reflect the absolute errors described in the text.

The following analysis was done with bias-corrected model data. Similar to monthly total precipitation, differences between absolute model error and absolute persistence error for monthly average temperature were lowest in winter and highest in summer when averaged by target month across lead zero to lead 11 (Fig. 21). Unlike for precipitation, however, the error difference was almost always above zero throughout the year, meaning that the absolute persistence error was smaller overall in the vast majority of months across the study area. When averaged by target month and lead-time (Fig. 22), negative error differences (i.e. smaller absolute model error than absolute persistence error) occurred only very rarely on very small number of grid cells along the Texas Gulf Coast (not shown) for forecasts for May (lead zero), July (leads two and three), and August (leads zero to two and four to 11), with error differences between -0.01 and -0.08 °C. Individual years had more pronounced differences than the overall lead and target year average (Fig. 23), suggesting a minimal advantage for the model forecast over a persistence forecast, mostly for small parts of the Texas Gulf Coast for May, June, and July. In all other monthlead averages the absolute persistence error was smaller.

Comparisons of lead-times leading to the same target month, as shown in Fig. 24 for November, April, and July forecasts at leads zero, four, eight, and 11 (Fig. 22 and Fig. 23) for all months, suggest that, unlike precipitation (Fig. 17), error differences in temperature are considerably dependent on lead-time. Differences during summer increased considerably with lead-time (Fig. 24), due to an overall greater absolute model error at longer lead-times (not shown).

Time series from 1985 to 2013 (Fig. 23) showed that the increase in error difference with lead-time varied considerably from year to year, and often increased with lead-time during summer and early fall (June to September), but to a much smaller extent in winter.



Fig. 21: Air temperature maps comparing absolute model and absolute persistence forecast errors for every target month, averaged for all lead-times. Red (blue) shaded areas indicate smaller persistence (model) error.



Fig. 22: Time series for temperature for January to December comparing averages of absolute model and absolute persistence forecast errors for lead 0 to 11. Graphs show the difference between absolute model and absolute persistence error. Values above (below) zero indicate smaller model (persistence) error.



Fig. 23: Temperature time series from 1985 to 2013 comparing averages of error differences for forecasts for January to December (lead zero and lead 11). Graphs show the difference between absolute model and absolute persistence error at lead zero (thin line) and lead 11 (thick line). Values above (below) zero indicate smaller model (persistence) error.

November, lead 0 Rage: 0.01 - 0.82 deg. C Mean: 0.45 deg. C

-4 -3.5 -3 -2.5 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4



0.5 1 1.5 2 2.5 3 3.5 4

November, lead 4

-4 -3.5 -3 -25 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4

April, lead 4 Range: 0.18-1.5 deg. C Mean: 1.07 deg. C

1 1.5 2 2.5 3 3.5 4

November, lead 8 Range: 0.17 - 1.37 deg. C Mean: 0.74 deg. C



4 -3.5 -3 -2.5 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4



-4 -3.5 -3 -2.5 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4

November, lead 11



-4 -3.5 -3 -2.5 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4

April, lead 11 Range: 0.2 - 2.17 deg. C Mean: 1.53 deg. C

-4 -3.5 -3 -2.5 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4

July, lead 0 July, lead 4 July, lead 8 July, lead 11 Mean: 3.29 deg. C Range: 0.02 - 1.25 deg. C : 0.62 deg. C Range: 0.01 - 3.93 deg. C Mean: 2.53 deg. C Range: 0.14 - 4.91 deg. C Range: 0.15 - 5.16 deg. C Mean: 3.52 deg. C ;-2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4 -2.5 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4 .5 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5 4 -2.5 -2 -1.5 -1 -0.5 0 0.5 1 1.5 2 2.5 3 3.5

Fig. 24: Air temperature maps comparing error differences between absolute model and absolute persistence errors for November, April, and July at leads 0, 4, 8, and 11. Red shaded areas indicate smaller persistence error.

3.3 Number of days per month with extreme precipitation

For this part of our analysis, we deviated from the survey results that drove this study, which asked for a measure of the "chances for extreme rainfall." Since we used a single model forecast (with 12 runs driven by different initial conditions) instead of an ensemble forecast with different individual models, the resulting autocorrelation between the 12 model runs would have created a false level of confidence in the forecast as compared to an ensemble that used models with independent model structure and physics. Therefore, we decided to study the number of days per month with extreme rainfall as a proxy to quantify the forecast accuracy of extreme rainfall. In our view, this measure could still provide agricultural producers with a useful long-term decision support product, assuming the accuracy was sufficient.

We used daily observations and daily model forecasts for total precipitation from 1985 to 2013. We also used the observational dataset (1980-2012) to calculate a daily persistence forecast for 1985 to 2013 in a way similar to the monthly persistence forecast, with the difference that the persistence forecast was the average of the same calendar day (not month) from the previous five years. We defined extremes thresholds using two indices from the Project to Develop Datasets for Indices of Climate Extremes (CLIMDEX), an initiative by the World Meteorological Organization (WMO) Expert Commission on Climate Change Detection and Indices (ETCCDI). CLIMDEX provides 27 extremes indices for various variables (<u>http://www.climdex.org/indices.html</u>). CLIMDEX extreme precipitation indices have recently been used in various studies analyzing extremes in weather and climate models over North America, e.g., Bennett and Walsh (2015), Mutiibwa et al. (2015), Curry et al. (2016), Werner and Cannon (2016), Schoof and Robeson (2016), or Sobie and Murdock (2017). We used the 95th and 99th percentile of days with precipitation of at least 1 mm as threshold for very wet days and extremely wet days respectively, as defined by CLIMDEX. Tab. 11 defines the observational and model datasets as well as CLIMDEX indices we used. Our bias correction caused a reduction of daily precipitation, which for some grid cells resulted in no precipitation above 1 mm, which would have rendered this analysis useless. Therefore, we conducted our analysis (as well as the analysis of dry days in section 3.4) on uncorrected data.

	Precipitation datasets
Observations	CPC Global Unified Gauge-Based Analysis of Daily Precipitation
Model	GFDL Forecast-oriented Low Ocean Resolution (FLOR) version B01
	CLIMDEX indices
R95pTOT	95th percentile of precipitation on days with \geq 1 mm precipitation
R99pTOT	99th percentile of precipitation on days with \geq 1 mm precipitation

Tab. 11: Precipitation datasets and extremes indices used.

Generally, both persistence and model forecasts underestimated high daily precipitation amounts and overestimated low daily precipitation amounts, but the geographic distribution of minima and maxima across the study region was similar to the monthly precipitation results. These phenomena of over- and under-prediction of low and extreme precipitation are a known problem in forecasting (Chakraborty 2010, Fernandez-Gonzalez et al. 2015, North et al. 2013, Rodwell et al. 2010, Zhou and Wang 2017) and had an effect on the thresholds of the 95th and 99th percentiles and their respective counts, shown in Fig. 25 (95th percentile) and Fig. 26 (99th percentile). The 95th percentile thresholds (Fig. 25, top) for observations, for example, ranged from 7.9 mm/day to 43.1 mm/day; the 95th percentile thresholds for model and persistence forecasts ranged from 3.3 mm/day to 8.5 mm/day and 3.4 mm/day to 15.5 mm/day

respectively. Therefore, applying the observational thresholds to the model data would have resulted in a vast under-estimation of extreme values in the model data. To account for the underestimation of extreme precipitation in model and persistence forecasts, individual thresholds for observations, model forecast, and persistence forecast were used, and the number of days per month exceeding the respective thresholds were normalized with the respective overall totals of values exceeding the 95th and 99th percentile per grid cell (Fig. 25 and Fig. 26, bottom). After normalizing, monthly totals of days exceeding the 95th and 99th percentile were calculated for observations as well as model and persistence forecasts, and model and persistence errors were calculated for each grid cell. After making these errors absolute, we calculated the difference in absolute model and persistence error. Values below zero meant smaller absolute model error, values above zero meant smaller absolute persistence error. Lastly, the errors were averaged by leadtime (0 to 11), target month (January to December), and year (1985 to 2013).

Overall, for the number of days per month exceeding the 95th (99th percentile), shorter leads showed greater model accuracy in most of the study area, while longer leads showed greater persistence accuracy in most of the study area (Fig. 27). More specifically, leads one to four (five) showed smaller absolute model errors in more than 50 percent of the study area, whereas leads five (six) to 11 showed smaller persistence errors in more than 50 percent of grid cells. An exception for both the thresholds is lead zero, in which the model forecast strong over-estimated the number of days with extreme precipitation, compared to leads one to four (five). The underlying cause for this is an abnormally high model error during the winter half year at lead zero for both the 95th and the 99th percentiles (see Fig. 28). The reason for this could be the low and relatively consistent winter precipitation, which might be captured more accurately by a persistence forecast

95th percentile threshold



Number of values above the 95th percentile



Fig. 25: 95th percentile thresholds (top) and the total number of daily values (1985-2013) that exceeded the threshold in each grid cell (bottom) for observations (left), model forecasts (center), and persistence forecast (right).

99th percentile threshold



Number of values above the 99th percentile



Fig. 26: 99th percentile thresholds (top) and the total number of daily values (1985-2013) that exceeded the threshold in each grid cell (bottom) for observations (left), model forecasts (center), and persistence forecast (right).



Fig. 27: Lead-time comparison of model and persistence error (across months January to December and years 1985 to 2013) for number of days per month above the 95th and 99th percentile. Bars indicate the portion of the study region with greater absolute model or persistence error.



Fig. 28: Detail on the lead zero bars in Fig. 27. Percentage of grid cells within the study region that show higher absolute model error or higher absolute persistence error for the number of days per month with precipitation above the 95th/99th percentile. The average percentages for model and persistence error in both charts roughly equal the respective lead zero bars in Fig. 27.

and was generally over-predicted by the model at lead zero due to initialization shock (see also footnote in section 3.1). On the other hand, precipitation extremes in summer caused by convective events might be less accurately captured by a persistence forecast, therefore a higher relative persistence error during summer compared to the model forecast. The transition from less than 50 percent model error to less than 50 percent persistence errors was gradual. Neither the geographic distribution in different lead-times nor the transition from short to long leads followed a spatial pattern, except an area with low model accuracy across most leads and target months were the Rocky Mountains in central Colorado, where the absolute model error was generally larger than the persistence error for both the number of days above the 95th and 99th percentile (Fig. 29).

A monthly comparison (across all lead-times and years 1985 to 2013), shown in Fig. 30, places the lowest model accuracy (highest persistence accuracy) between November and April, with a gradual increase (decrease) in accuracy from April to October. This pattern is evident for most lead-times (not shown).

A year-by-year comparison (across all months and lead-times), displayed in Fig. 31, shows that model (persistence) error was greater in more than 50 percent of grid cells for 17 (12) years. It is worth noting that in 2011, a year with severe summer drought in much of the Southern Great Plains, on average 84 (82) percent of grid cells showed smaller absolute model error than persistence error for days above the 95th (99th) percentile, more than in any other year, especially during the summer months (not shown) when the drought had the greatest spatial extent and was most severe in the Southern Great Plains (Svoboda et al. 2002, Southern Climate Impacts Planning Program 2018). The reason for this is likely because the five previous years (which went into the persistence forecast) did not have such unseasonal conditions and were thus unable to represent the drought year.

Meanwhile, 2011 was a La Niña year, with the Ocean Niño Index (ONI), a three month average of sea surface temperatures in the Niño 3.4 region, smaller than -0.5 in nine of 12 month (http://origin.cpc.ncep.noaa.gov/products/analysis_monitoring/ ensostuff/ONI_v5.php). The year also had a stronger-than-average correlation between ONI and absolute model error for days per month exceeding the 95th percentile (however, not for the 99th percentile), as shown in Fig. 32. Therefore, it could be assumed that 2011 was more predictable than other years, and while the model forecast for 2011 also showed high errors (not shown), the year might have been somewhat more predictable by model compared to persistence.



Fig. 29: Distribution of differences between the absolute model error and absolute persistence error in estimating the number of days per month with precipitation above the 95th (left) and 99th (right) percentile of daily precipitation. Purple (green) grid cells represent smaller absolute model (persistence) errors.



Fig. 30: Monthly comparison of model and persistence error (average of leads zero to 11 and years 1985 to 2013) for number of days per month above the 95th and 99th percentile. Bars indicate the portion of the study region with greater absolute model or persistence error.



Fig. 31: Yearly comparison of model and persistence error (across leads zero to 11 and months of January to December) for number of days per month above the 95th and 99th percentile. Bars indicate the portion of the study region with greater absolute model or persistence error.

3.4 Number of dry days per month

For this section, we compared model and persistence forecasts in their ability to correctly predict the number of days per month with less than 1 mm of precipitation. We chose this threshold because we have also used it in the previous section as threshold to define wet days (days with at least 1 mm daily precipitation) in accordance with WMO standards, and it is also used as threshold in other ETCCDI indices. As in the previous analysis, we decided to deviate from the extension survey informing this study, for feasibility and practicality reasons. The survey requested a forecast for the number of consecutive days with no precipitation per month, which means the forecast skill received a penalty for forecasting the incorrect amount of precipitation *and* for forecasting precipitation on the wrong day. This test was difficult, even in the short term, and it appeared to be information that farmers would unlikely use for long-term decisions because of uncertainty in other decision factors, such as market prices (see Klockow et al.



Fig. 32: Pearson correlation between ENSO 3.4 index (3-month average deviation from the long-term normal) and monthly absolute model error for the number of days above the 95th and 99th percentile for lead zero to 11. Solid lines show the correlation between ENSO and all absolute errors. The dashed blue (red) line shows the correlation for 2011 for the absolute error for days/month exceeding the 95th (99th) percentile for lead zero to 11.

2010 for details). Therefore, as a more feasible and practical proxy to the forecast element requested in the survey, we analyzed the total number of dry days per month.

Both model and persistence forecasts considerably under-predicted the average number of dry days per month observed between 1985 and 2013 in our study area (Fig.



Fig. 33: Average number of dry days per month from 1985 to 2013 for observed daily precipitation (top left) and predicted by persistence forecast (top center) and model forecast (top right). Range and Averages state the average number of days per month on the respective maps. The graphs (bottom) show the average number of dry days per calendar month, for observations (black double line), persistence (red dashed line), and leads zero to 11 of the model forecast (lead zero to five: blue solid lines, lead six to 11: dotted grey lines, lead zero to 11 average: blue dashed line).

33, top). On average, 23.3 days per month recorded precipitation of less than 1 mm. The persistence forecast estimated an average of 16 days per month, and the model forecast averaged 11.3 days per month. As a result, the persistence forecast under-predicted on average by 7.3 days per month, and the model forecast under-predicted on average by 12 days per month. Note: These errors reflect the average raw errors, meaning averages of over- and under-prediction. For the remainder of this section we analyzed averages of absolute errors, meaning the magnitude but not the direction. Therefore the following average errors will not be exactly the same as the ones just mentioned.

The observed spatial distribution of minima and maxima in average monthly dry days was maintained in persistence and model forecasts, as shown in Fig. 33 (top), with minima in all three cases located in western Colorado as well as eastern Oklahoma and Texas, and maxima located in a band from western Texas to eastern Colorado. The temporal distribution of minima and maxima throughout the year, shown in Fig. 33 (bottom), was different between model forecast, observations, and persistence forecast. While observations and the persistence forecast have minima in June and maxima in December and January, the model forecast predicts local maxima during the summer (lead zero and lead one even have their highest overall averages in July) and minima in May and June for lead zero to lead three and lead four to lead 11, respectively. Despite the temporal mismatch between model forecast and observations, in parts of the study area for several months and lead-times the model forecast had smaller absolute errors than the persistence forecast. Fig. 34 compares the average absolute errors of model and persistence forecasts of dry days per calendar month. While the absolute model error generally increases with lead-time, it is smaller than absolute persistence error in at least one, and as many as six calendar months. For example, between May and October at lead zero (Fig. 35, top left), 73 to 90 percent of the study area (mainly the entire study area except for parts of Colorado) had smaller absolute model errors compared to persistence errors. This percentage decreased with longer lead-times; however, certain regions maintained the smaller absolute model errors through lead 11. For example, Fig. 36 shows that in the month of August most of the study region had smaller absolute model error in more than 50 percent of the area at shorter leads zero to four, and even up to lead 11, the



Fig. 34: Average absolute persistence and model forecast errors. The dashed red line shows the average persistence error, the blue shaded solid lines and grey shaded dotted lines show the model error for leads zero to five and six to 11 respectively, the blue dashed line shows the lead zero to 11 average.



Fig. 35: The percentage of study area with smaller absolute persistence error (green) and smaller absolute model error (red) for forecasts at lead zero, four, seven, and 11.



Fig. 36: The spatial distribution of areas with smaller absolute model errors (green) and areas with smaller absolute persistence errors (red) for leads zero, three, seven and 11 forecasts for forecasts for the month of August. Numbers above the maps show the percentage of grid cells in green (left) and red (right). Sums different from 100 percent are due to rounding.

Texas Gulf Coast, the absolute model error remained smaller than the absolute persistence error. This spatial distribution and temporal change was similar in June, July, and September (not shown).

Although model and persistence forecasts performed better or worse relative to each other in some months and geographic areas, neither of them performed particularly well compared to observations, as shown in Fig. 33 and Fig. 34 and described earlier. The absolute persistence error, averaged for all calendar months, was 7.8 days per month; the average absolute model error, averaged for all calendar months at lead zero, was 7.3 days per month and nearly doubled to 14 days per month at lead 11 (not shown). The average absolute model error for May to October at lead zero was 6.4 days per month (14.9 days per month at lead 11). Fig. 37 compares for each calendar month and lead-time how many percent of observed dry days per month were predicted by model and persistence forecast. The persistence forecast predicted, on average, 68 percent of observed dry days per month. The model forecast at lead zero predicted 75.3 percent of the observed number of dry days per month (39.7 percent at lead 11). From May to October, the model forecast predicted, on average, 82.7 percent of observed dry days per month at lead zero (33.1 percent at lead 11). The lowest percentages by both persistence and model forecast occurred between April and June; the lowest persistence forecast was 52.2 percent (June).



Fig. 37: Percentage of the numbers of dry days/month as predicted by persistence and model forecast. The dashed red line shows the average persistence error, the blue shaded solid lines and the grey shaded dotted lines show the model error for leads zero to five and six to 11 respectively. The black dash-dotted line shows the model lead zero to 11 average. Note: The underlying absolute numbers are based on averages of the raw data (the same data that was used in Fig. 27 and Fig. 28); therefore, the percentages here do not match the absolute forecast errors of the respective month (and lead-time).

The lowest model forecast ranged between 60.5 percent at lead zero (occurred in April) to 14.2 percent at lead 11 (June).

4. Summary and discussion

In the previous section we compared the accuracy of model and persistence forecasts related to four forecast elements: total monthly precipitation, average monthly air temperature, number of days per month with extreme precipitation, and number of dry days per month. The thought behind this analysis was to compare model accuracy against an alternative forecast product (a persistence forecast) that farmers might create from publicly available weather observations and use in lieu of an actual model forecast. In this section, we summarize our findings and put them in context by reviewing and comparing them to existing literature.

Average total monthly precipitation was more accurately forecasted by the persistence forecast in most months and across most of the study area, especially in summer. However, forecasts for November to February for all lead-times across central and western Texas and Oklahoma, showed greater accuracy in the model forecast. Accuracy of average precipitation had considerable seasonality and depended overall more on the target month rather than the forecast lead-time. This disagreed with previous studies by Schneider and Garbrecht (2006), who studied probabilistic seasonal outlooks by CPC and found that forecast skill for wetter- and dryer-than-average forecasts existed mainly along the Texas coast and dropped considerably after lead zero, in addition to longer lead-times having a strong tendency towards the long-term average. Our results found that no model forecast improvement over time, which agreed with Krakauer et al. (2013), with the caveat that they studied CPC seasonal outlooks that were created with a different model than the GFDL FLOR. The seasonal character of precipitation accuracy was also found by van den Dool (1994) and Peng et al. (2013), albeit also studying CPC's seasonal outlooks.

Accuracy of average monthly temperature, unlike precipitation, was very dependent on lead-time. For virtually all months and lead-times, but especially in summer at longer lead-times, model accuracy was lower than persistence accuracy. This result also contradicts Schneider and Garbrecht (2006), who found that forecast skill for warmerthan-average forecasts in our study area existed throughout all lead-times, but with a tendency of forecasts closer to climatology at longer lead-times. Cooler-than-average forecasts, meanwhile, rarely differed from climatology in their study. These results also contrast with other studies that generally showed that seasonal temperature forecasts had higher skill than seasonal precipitation forecasts (Schneider and Garbrecht 2003b, Schneider and Garbrecht 2006, Kerr 2008).

Analyzing daily extremes, we found that while dry days were underestimated by model and persistence forecast, our data suggested that low precipitation amounts were actually overestimated and high precipitation amounts were underestimated by both model and persistence forecasts. We attributed this result to the fact that individual model and persistence forecasts were averages of several datasets (preceding years for the persistence forecast, 12 model runs for the model forecast), which meant that single extreme values (both dry and wet) present in one dataset could be smoothed with data from other datasets. Thus, low precipitation values could be overestimated and high precipitation could be underestimated, which is a problem also discussed in other studies, with similar conclusions (Knutti et al. 2010, Barnston and Mason 2011, Huang and Gao 2017). The number of days with extreme precipitation was estimated more accurately by the persistence forecast at longer lead-times and lead zero across most of the study area during most of the year. However, the model forecast was more accurate in most of the region during summer at shorter lead-times, with the exception of lead 0, suggesting the highest model accuracy at these lead-times.

The number of dry days per month was more accurately forecasted by the model than by the persistence forecast in summer across most of the study area at short lead-times, including lead zero, and along the Texas Gulf coast at all lead-times in summer. However, both persistence and model forecasts vastly underestimated the number of dry days per month. The persistence forecast predicted between 52 and 84 percent, the model forecast (lead zero to 11 average) predicted between 24 and 65 percent of the observed dry days per month. Interestingly, forecasting dry days and extreme precipitation days, the model was most accurate compared to persistence in months when it was least accurate in predicting average precipitation.

Using a single-model approach over a model ensemble approach to create tailored forecasts (or any forecasts for that matter) has advantages and disadvantages. Computationally, running a single model reduces cost and time over running multiple forecast models; however, model ensembles are generally seen as more skillful than single models in predicting average temperature and precipitation (Knutti et al. 2010, Stockdale et al. 2010, DelSole and Tippett 2014, Kirtman et al. 2014). For the purpose of cost-effective yet skillful seasonal forecasts, further research should be conducted in model comparison.

5. Conclusion

Our analysis answered our initial research question, whether or not the model forecast is more accurate than a persistence forecast for forecast elements requested by winter wheat producers in our study area. In closing, we would like to leave the reader with two considerations: (1) Although the model forecast is more accurate than persistence at times, it might not be accurate enough to serve actual decision making, and future studies should explore ways and thresholds to determine this answer. (2) Depending on time and location, a model forecast might not always be the best basis for decision making, and the decision maker may instead (also) consider a persistence forecast, as it might be able to better inform about future conditions. Future work should include improving forecast models' ability to predict extremes and other variables relevant to agricultural producers. Studies should also explore ways to assess and communicate uncertainty of the forecast, such that user can determine how much to trust and whether to use the forecast.

6. Acknowledgements

We thank Dr. Esther Mullens for help writing the NCL scripts for our analysis, and to Drs. Derek Rosendahl and Adrienne Wootten for advice on our analysis. We would also like to thank Mark Stacy, Tyler Pearson, and the support staff at the University of Oklahoma's Supercomputing Center for Research and Education (OSCER) for troubleshooting help during our analysis, and we thank the South Central Climate Science Center for providing computing power and data storage for this study.

7. Conflict of interest statement

The authors have no conflicts of interest to declare for this study.

Chapter 5 - Summary and Conclusion

1. Summary

Extreme weather and climate, such as drought, heat, or extreme rainfall, can have a destructive impact on agricultural productivity, with severe consequences. In recent years world events, such as the famine at the Horn of Africa or the civil war in Syria, have shown that given the "right" (or rather, wrong) political, societal, and economic circumstances, failing agricultural systems can contribute to economic disruption, food insecurity, famine, and even conflicts. In the U.S., the 2011 to 2014 drought in the Great Plains, the Midwest, and California has demonstrated how vulnerable agricultural systems are even in highly developed countries.

For several decades in the U.S., forecast information has been issued by the National Weather Service following a loading-dock approach (Cash et al. 2006), issuing forecast information without user input or feedback and without analyzing who is using the forecasts. Without involvement from users, such as forest managers, policy makers, city planners, or agricultural producers, and without coordinated planning and incorporation of user needs these forecasts can leave recipients uninformed or even misinformed, potentially causing more problem than they solve (Cash et al. 2006, Meadow et al. 2015).

The research conducted in this dissertation examined the seasonal forecast needs of producers of winter wheat and compared the accuracy of a high-resolution seasonal climate forecast model in providing forecasts elements as requested in the survey for the Southern Great Plains. In collaboration with the agricultural community, this research intended to answer two questions:

1. How can seasonal climate forecasts be tailored to serve the needs of winter wheat growers in the south-central United States?

2. Can existing seasonal forecast models provide meteorological variables as requested by winter wheat farmers with better skill than a persistence forecast?

Survey data collected from 109 surveys submitted by cooperative extension agents working in Colorado, Kansas, Oklahoma, or Texas highlighted that:

- Decision planning occurred about zero to 2.5 month before the respective practice (e.g., planting, harvesting) was carried out, suggesting a strong seasonality and timing closely tied to the timing of the practice.
- Drought and extreme rainfall are the overall most relevant weather threats with regard to long-term decision timing. However, relevance of threats differs from one decision to another. For example, while drought, extreme rainfall, and heat most affect planning for planting time, extreme rainfall, wind/storm, and hail are most closely tied to the decision about when to harvest, and drought and heat play a less important role.
- Overall, forecast elements related to precipitation ranked higher in importance than forecast elements related to temperature, which is consistent with existing literature. In particular, average precipitation and consecutive days without precipitation ranked highest and second highest in every month and overall. Deviation from average precipitation and chances for extreme precipitation ranked third and fourth overall. Average temperature followed in fifth place and was followed by six other temperature-related forecast elements.

Of all ranked forecast elements, only the first, third, fifth, and seventh (average precipitation, deviation from average precipitation, average temperature, and deviation from average temperature) are currently provided through seasonal forecasts issued by the NWS for the U.S.

Unexpected survey analysis results included the required forecast lead-time of zero to 2.5 months, and the low ranking of growing degree days, a decision-support tool specifically developed for agricultural and horticultural users. This suggested that on a seasonal timescale, winter wheat producers do not require this tool as much as they do on shorter forecast timescales, as found by Haigh et al. (2015) for corn producers in the U.S. Midwest.

Following the survey, a quantitative analysis was conducted to compare a highresolution seasonal climate forecast model with a persistence forecast regarding accuracy of monthly forecasts for four of the five forecast elements ranked highest in the survey: average precipitation, average temperature, chances for extreme rainfall, and consecutive days without rainfall. The model that was chosen is developed and operated by the NOAA Geophysical Fluid Dynamics Laboratory (GFDL) and produces daily and monthly forecasts on a 50 by 50 km grid with lead-times from zero to 11 months, zero being the month the forecast is issued. The model consists of 12 model runs with differing initial conditions, which were averaged (unweighted) for the analysis. In the process, two of the elements were changed for reasons of feasibility, from chances for extreme precipitation to the number of days per month with extreme precipitation, and from consecutive days without precipitation to the number of days per month without precipitation. The persistence forecast for a particular calendar month (day) was calculated by averaging the observational values of the same calendar month (day) from the preceding five years. Absolute forecast and persistence errors were created using monthly air temperature observations from the Global Historical Climatology Network as well as daily and monthly precipitation observations from CPC and NOAA, respectively. Finally, these absolute errors were compared for each calendar month and lead-time. The following results are worth mentioning:

- On average, average precipitation was forecasted more accurately by the model forecast between November and February for all lead-times in large parts of the study area, and more accurately by the persistence forecast between June and September nearly throughout the study area. October and March to May were transition months. This dependence of model accuracy on season rather than lead-time is consistent with existing literature.
- On average, monthly air temperature was forecasted more accurately by the persistence forecast in almost every month and lead-time, and in particular during summer at longer lead-times.
- Model accuracy of monthly temperature and precipitation forecasts for summer has likely been decreased due to bias correction in combination with the fact that the model used was not flux-adjusted.
- On average, the number of days with extreme precipitation was estimated more accurately by the persistence forecast at longer lead-times and lead zero across most of the study area during most of the year. However, the model forecast was more

accurate in most of the region during summer at shorter lead-times, with the exception of lead 0.

- On average, the number of dry days per month was forecasted more accurately by the model across most of the study area between May and October at lead-times of one to up to four. Along the Texas gulf coast from June to September, the model forecast was more accurate at all lead-times. Generally though, longer lead-times and forecasts for November to April were more accurately forecasted in most of the study region by the persistence forecast.
- Overall, both persistence and model forecasts greatly underestimated the number of days per month with extreme precipitation and underestimated the number of dry days per month. This is presumably caused by the described averaging of observations and model runs.

Unexpected was the low absolute model accuracy of average monthly temperature relative to absolute persistence accuracy, which, unlike the comparison of forecasts for total monthly precipitation, was lower almost throughout the year and across the entire study area for every lead-time. This was surprising considering that temperature forecasts are generally more accurate than precipitation forecasts. Unexpected were findings that model forecasts for precipitation averages were more often more accurate than persistence forecasts during winter, but model forecasts for precipitation extremes were more accurate than persistence forecasts in summer.

This research focused on one crop type instead of multiple crop types or crop production in general in order to taking into account the different decision-making processes that distinguish winter wheat production from other crop types, for example, corn, cotton, or soybean production. Winter wheat is the dominant crop type by planted acreage in the study region (Texas, Oklahoma, Kansas, and Colorado), grown on 21.1 million acres (in 2016), twice the area of the second largest crop, corn (Han et al. 2012). As a strain of wheat, it also contributes 71 percent to the U.S. wheat harvest (USDA 2012). Wheat itself is the third largest crop by harvested acreage in the U.S., after corn and soybean. Therefore, because of its overall contribution to the national wheat harvest and its role in the study region, compared to other crops, it was chosen as focus crop for this research.

The goal of this research was to help forecasters provide better decision support for agricultural producers generally and winter wheat producers in particular, to provide model developers with insights into model performance regarding specific user needs, and to highlight the advantages of and continued need for collaborative, interdisciplinary efforts towards a better understanding of decision-making processes in agricultural production systems.

2. Conclusion

This research, broadly speaking, illustrated some of the benefits of co-produced research over (previous) research that was produced in "academic silos" with little or no interaction with and/or input from users. It showed ways in which seasonal climate forecasts can be improved to serve as better decision tools in winter wheat production, and it highlighted shortcomings of a current seasonal climate forecast model in producing such tailored forecasts. The results, produced in a collaborative, cross-disciplinary way, add to the existing body of knowledge by highlighting decision processes and forecast

needs for a specific crop type. They pointed out several forecast elements relevant to winter wheat producers that could currently be provided by seasonal forecast models. However, due to the complex nature of farm decision making, it should be emphasized that even if a forecast was perfect producers may choose not to act upon them due to other limitations, such as risk, market conditions, or other climatic factors.

The forecast comparison highlighted three aspects. Discounting model forecast data out of skepticism in favor of observation-based data, such as seasonal persistence forecasts (Changnon et al. 1988, Sonka et al. 1992), or vice versa, might put decision makers at a disadvantage, because they may rely on more inaccurate information. However, while both model and persistence forecasts were more or less accurate relative to each other at certain times across the study region, both were also very inaccurate in certain instances to begin with. Lastly, past research showed that model ensembles are in many cases more accurate than single models due to the smoothing effect of averaging several datasets which reduces overall errors. The analysis of extremes, however, suggested that this smoothing can cause considerable misrepresentation of extreme values, such as dry days or extreme precipitation days.

This work is an important stepping stone in lifting research out of the metaphorical silos of academia and closer to the literal silos of agricultural production. Scientific research often takes place isolated from the real world and is often taught that way. While not all research needs practical context, real-world urgency, such as the increasing impacts of climate variability and climate change on various aspects of everyday life, can make it necessary to go a step further and explore ways in which scientific knowledge can have real-world use. While the research presented here is likely incomplete and in only a few years outdated, it is a step towards helping society adapt to and mitigate climate
variability and change by producing knowledge and developing decision tools that can help secure wellbeing and prosperity, and by advancing ways of collaboration between scientists and decision makers.

3. Future work

Identifying user needs has been done for other agricultural commodities in different parts of the U.S. and internationally (Unganai et al. 2013, Haigh et al. 2015, Takle et al. 2014, Goddard et al. 2010, Klemm and McPherson 2017), but more work is necessary to connect these puzzle pieces to a more complete picture and to provide guidelines for best practices. The development and testing of forecast models can benefit from these userinspired results, too, for example, by making model evaluation and comparison more practically relevant. The value of scientist-stakeholder relationships and science coproduction as it exists, for example, within the Climate Science Center network, should to be promoted among and shared with other agencies that are not following these practices yet or are struggling to adopt them. Qualitative and quantitative metrics to evaluate the success of collaborative projects should be developed to accurately, truthfully, and comprehensively describe the value of this research to funding agencies, policymakers, natural and cultural resource managers, and researchers (Wall, McNie and Garfin 2017). Research frameworks also need to be developed, improved upon, and standardized to assure compatible and quality results (Meadow et al. 2015, Buizer, Jacobs and Cash 2016). Last but certainly not least, college coursework is necessary to train future researchers in understanding, appreciating, and incorporating scientific findings from multiple disciplines (Evans et al. 2015, Hill et al. 2014).

Bibliography

- Adams, R. M., L. L. Houston, B. A. McCarl, M. Tiscareño L, J. Matus G & R. F. Weiher (2003) The benefits to Mexican agriculture of an El Niño-southern oscillation (ENSO) early warning system. Agricultural and Forest Meteorology, 115, 183-194.
- Agrawala, S., K. Broad & D. H. Guston (2001) Integrating climate forecasts and societal decision making: Challenges to an emergent boundary organization. *Science Technology & Human Values*, 26, 454-477.
- Alessandri, A., A. Borrelli, A. Navarra, A. Arribas, M. Déqué, P. Rogel & A. Weisheimer (2011) Evaluation of Probabilistic Quality and Value of the ENSEMBLES Multimodel Seasonal Forecasts: Comparison with DEMETER. *Monthly Weather Review*, 139, 581-607.
- Allen, E. & J. C. Stephens. 2016. Enhancing the Usability of Climate Information and Models Through Stakeholder Engagement. In Information, Models, and Sustainability: Policy Informatics in the Age of Big Data and Open Government, eds. J. Zhang, F. L. Luna-Reyes, A. T. Pardo & S. D. Sayogo, 121-135. Cham: Springer International Publishing.
- Babbie, E. R. 2014. The Basics of Social Research.
- Bales, R. C., D. M. Liverman & B. J. Morehouse (2004) Integrated Assessment as a Step Toward Reducing Climate Vulnerability in the Southwestern United States. *Bulletin of the American Meteorological Society*, 85, 1727-1734.
- Balmaseda, M. A., M. K. Davey & D. L. T. Anderson (1995) Decadal and Seasonal Dependence of ENSO Prediction Skill. *Journal of Climate*, 8, 2705-2715.
- Barham, E. H. B., J. R. C. Robinson, J. W. Richardson & M. E. Rister (2011) Mitigating Cotton Revenue Risk Through Irrigation, Insurance, and Hedging. *Journal of Agricultural and Applied Economics*, 43, 529-540.
- Barnett, T. P., M. Latif, N. E. Graham, M. Flugel, S. Pazan & W. White (1993) ENSO AND ENSO-RELATED PREDICTABILITY .1. PREDICTION OF EQUATORIAL PACIFIC SEA-SURFACE TEMPERATURE WITH A HYBRID COUPLED OCEAN-ATMOSPHERE MODEL. *Journal of Climate*, 6, 1545-1566.
- Barnston, A. G., Y. He & D. A. Unger (2000) A Forecast Product that Maximizes Utility for State-of-the-Art Seasonal Climate Prediction. *Bulletin of the American Meteorological Society*, 81, 1271-1279.
- Barnston, A. G. & S. J. Mason (2011) Evaluation of IRI's Seasonal Climate Forecasts for the Extreme 15% Tails. Weather and Forecasting, 26, 545-554.
- Barnston, A. G., S. J. Mason, L. Goddard, D. G. Dewitt & S. E. Zebiak (2003) Multimodel Ensembling in Seasonal Climate Forecasting at IRI. Bulletin of the American Meteorological Society, 84, 1783-1796.
- Barnston, A. G., H. M. van den Dool, D. R. Rodenhuis, C. R. Ropelewski, V. E. Kousky, E. A. O'Lenic, R. E. Livezey, S. E. Zebiak, M. A. Cane, T. P. Barnett, N. E. Graham, M. Ji & A. Leetmaa (1994) Long-Lead Seasonal Forecasts—Where Do We Stand? *Bulletin of the American Meteorological Society*, 75, 2097-2114.
- Bennett, K. E. & J. E. Walsh (2015) Spatial and temporal changes in indices of extreme precipitation and temperature for Alaska. *International Journal of Climatology*, 35, 1434-1452.
- Beraki, A. F., D. G. DeWitt, W. A. Landman & C. Olivier (2014) Dynamical Seasonal Climate Prediction Using an Ocean-Atmosphere Coupled Climate Model Developed in Partnership between South Africa and the IRI. *Journal of Climate*, 27, 1719-1741.
- Bonhomme, R. (2000) Bases and limits to using 'degree.day' units. European Journal of Agronomy, 13, 1-10.

- Boulanger, J.-P. & O. Penalba (2010) Assessment of climate information needs in the Argentinean Agro-business sector. *Climatic Change*, 98, 551-563.
- Breuer, N. E., S. Adhikarim, R. Brown-Salazar, J. A. Clavijo, H. N. HansPetersen, N. C. Kawa, R. Patarasuk & P. Hildebrand. 2008. Extension Agent Perspectives of Climate, Seasonal Climate Forecasts, and the AgClimate Decision Support System. In Southeast Climate Consortium Technical Report Series. Southeast Climate Consortium SECC.
- Breuer, N. E., D. Zierden, J. G. Bellow, J. Paz, V. E. Cabrera, A. Garcia y Garcia, K. T. Ingram, U. Hatch, G. Hoogenboom, J. W. Jones & J. J. O'Brien (2006) AgClimate: A climate forecast information system for agricultural risk management in the southeastern USA. *Computers and Electronics in Agriculture*, 53, 13-27.
- Briley, L., D. Brown & S. E. Kalafatis (2015) Overcoming barriers during the co-production of climate information for decision-making. *Climate Risk Management.*
- Buizer, J., K. Jacobs & D. Cash (2016) Making short-term climate forecasts useful: Linking science and action. Proceedings of the National Academy of Sciences, 113, 4597-4602.
- Cabrera, V. E., N. E. Breuer, J. G. Bellow & C. W. Fraisse. 2006. Extension Agent Knowledge and Perceptions of Seasonal Climate Forecasts in Florida. In Southeast Climate Consortium Technical Report Series, SECC Technical Report 06-001. Gainsville, FL: Southeast Climate Consortium.
- Cabrera, V. E., D. Letson & G. Podestá (2007) The value of climate information when farm programs matter. *Agricultural Systems*, 93, 25-42.

Bontkes, T. S. & M. Wopereis. 2003. *Decision support tools for smallholder agriculture in Sub-Saharan Africa : a practical guide*. Muscle Shoals, Ala., U.S.A.; Wageningen, Netherlands: IFDC; ACP-EU Technical Centere for Agricultural and Rural Cooperation.

- Carberry, P., G. Hammer, H. Meinke & M. Bange. 2000. The Potential Value of Seasonal Climate Forecasting in Managing Cropping Systems. In *Applications of Seasonal Climate Forecasting in Agricultural and Natural Ecosystems*, eds. G. L. Hammer, N. Nicholls & C. Mitchell, 167-181. Springer Netherlands.
- Cargill. 2015. Cargill to sell Plainview, Texas, beef plant site idled in 2013.
- Cash, D. W., J. C. Borck & A. G. Patt (2006) Countering the Loading-Dock Approach to Linking Science and Decision Making: Comparative Analysis of El Niño/Southern Oscillation (ENSO) Forecasting Systems. *Science, Technology & Human Values*, 31, 465-494.
- Chakraborty, A. (2010) The Skill of ECMWF Medium-Range Forecasts during the Year of Tropical Convection 2008. *Monthly Weather Review*, 138, 3787-3805.
- Changnon, S. A. (2004) Changing Uses of Climate Predictions in Agriculture: Implications for Prediction Research, Providers, and Users. *Weather and Forecasting*, 19, 606-613.
- --- (2007) The Past and Future of Climate-Related Services in the United States. Journal of Service Climatology, 1, 1-7.
- Changnon, S. A., S. T. Sonka & S. Hofing (1988) Assessing Climate Information Use in Agribusiness. Part I: Actual and Potential Use and Impediments to Usage. *Journal of Climate*, 1, 757-765.
- Chen, M., W. Shi, P. Xie, V. B. S. Silva, V. E. Kousky, R. Wayne Higgins & J. E. Janowiak (2008) Assessing objective techniques for gauge-based analyses of global daily precipitation. *Journal of Geophysical Research: Atmospheres*, 113, n/a-n/a.
- Chen, M., P. Xie, J. E. Janowiak & P. A. Arkin (2002) Global Land Precipitation: A 50-yr Monthly Analysis Based on Gauge Observations. *Journal of Hydrometeorology*, **3**, 249-266.
- Choi, H. S., U. A. Schneider, L. Rasche, J. Cui, E. Schmid & H. Held (2015) Potential effects of perfect seasonal climate forecasting on agricultural markets, welfare and land use: A case study of Spain. Agricultural Systems, 133, 177-189.
- Coelho, C. A. S. & S. M. S. Costa (2010) Challenges for integrating seasonal climate forecasts in user applications. *Current Opinion in Environmental Sustainability*, 2, 317-325.
- Colorado State University. 2010. Wheat Production and Pest Management for the Great Plains Region. Colorado State University Extension, Bioagricultural Sciences and Pest Management.
- Crane, T. A., C. Roncoli, J. Paz, N. Breuer, K. Broad, K. T. Ingram & G. Hoogenboom (2010) Forecast Skill and Farmers' Skills: Seasonal Climate Forecasts and Agricultural Risk Management in the Southeastern United States. Weather, Climate, and Society, 2, 44-59.
- Crochemore, L., M.-H. Ramos & F. Pappenberger (2016) Bias correcting precipitation forecasts to improve the skill of seasonal streamflow forecasts. *Hydrology and Earth System Sciences*, 20, 3601-3618.
- Curry, C. L., B. Tencer, K. Whan, A. J. Weaver, M. Giguere & E. Wiebe (2016) Searching for Added Value in Simulating Climate Extremes with a High-Resolution Regional Climate Model over Western Canada. *Atmosphere-Ocean*, 54, 364-384.
- Davies, J. R., D. P. Rowell & C. K. Folland (1997) North Atlantic and European seasonal predictability using an ensemble of multidecadal atmospheric GCM simulations. *International Journal of Climatology*, 17, 1263-1284.
- DelSole, T. & M. K. Tippett (2014) Comparing Forecast Skill. Monthly Weather Review, 142, 4658-4678.
- Delworth, T. L., A. J. Broccoli, A. Rosati, R. J. Stouffer, V. Balaji, J. A. Beesley, W. F. Cooke, K. W. Dixon, J. Dunne, K. A. Dunne, J. W. Durachta, K. L. Findell, P. Ginoux, A. Gnanadesikan, C. T. Gordon, S. M. Griffies, R. Gudgel, M. J. Harrison, I. M. Held, R. S. Hemler, L. W. Horowitz, S. A. Klein, T. R. Knutson, P. J. Kushner, A. R. Langenhorst, H.-C. Lee, S.-J. Lin, J. Lu, S. L. Malyshev, P. C. D. Milly, V. Ramaswamy, J. Russell, M. D. Schwarzkopf, E. Shevliakova, J. J. Sirutis, M. J. Spelman, W. F. Stern, M. Winton, A. T. Wittenberg, B. Wyman, F. Zeng & R. Zhang (2006) GFDL's CM2 Global Coupled Climate Models. Part I: Formulation and Simulation Characteristics. *Journal of Climate*, 19, 643-674.
- Delworth, T. L., A. Rosati, W. Anderson, A. J. Adcroft, V. Balaji, R. Benson, K. Dixon, S. M. Griffies, H. C. Lee, R. C. Pacanowski, G. A. Vecchi, A. T. Wittenberg, F. R. Zeng & R. Zhang (2012) Simulated Climate and Climate Change in the GFDL CM2.5 High-Resolution Coupled Climate Model. *Journal of Climate*, 25, 2755-2781.
- Dilley, M. (2000) Reducing Vulnerability to Climate Variability in Southern Africa: The Growing Role of Climate Information. *Climatic Change*, 45, 63-73.
- Dillman, D. A., J. D. Smyth & L. M. Christian. 2014. Internet, phone, mail, and mixed-mode surveys: the tailored design method. John Wiley & Sons.
- Diro, G. T. (2015) Skill and economic benefits of dynamical downscaling of ECMWF ENSEMBLE seasonal forecast over southern Africa with RegCM4. *International Journal of Climatology*, n/a-n/a.
- DOI. 2009. Secretarial Order 3289. U.S. Department of the Interior, Washington, D.C.
- Evans, J., R. Jones, A. Karvonen, L. Millard & J. Wendler (2015) Living labs and co-production: university campuses as platforms for sustainability science. *Current Opinion in Environmental Sustainability*, 16, 1-6.
- Fan, Y. & H. van den Dool (2008) A global monthly land surface air temperature analysis for 1948–present. Journal of Geophysical Research: Atmospheres, 113, n/a-n/a.
- FAO. 2014. FAOSTAT.

- Fernandez-Gonzalez, S., F. Valero, J. L. Sanchez, E. Gascon, L. Lopez, E. Garcia-Ortega & A. Merino (2015) Analysis of a seeder-feeder and freezing drizzle event. *Journal of Geophysical Research-Atmospheres*, 120, 3984-3999.
- Fiebrich, C. A. (2009) History of surface weather observations in the United States. Earth-Science Reviews, 93, 77-84.
- Flato, G., J. Marotzke, B. Abiodun, P. Braconnot, S. C. Chou, W. D. Collins, P. M. Cox, F. Driouech, S. Emori, V. Eyring, C. Forest, P. Gleckler, E. Guilyardi, C. Jakob, V. Kattsov, C. Reason & M. Rummukainen. 2013. Evaluation of Climate Models. In *Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*, eds. T. F. Stocker, D. Qin, G.-K. Plattner, M. M. B. Tignor, S. K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex & P. M. Midgley, 741-866. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Fraisse, C. W., N. E. Breuer, D. Zierden, J. G. Bellow, J. Paz, V. E. Cabrera, A. Garcia y Garcia, K. T. Ingram, U. Hatch, G. Hoogenboom, J. W. Jones & J. J. O'Brien (2006) AgClimate: A climate forecast information system for agricultural risk management in the southeastern USA. *Computers and Electronics in Agriculture*, 53, 13-27.
- Frisvold, G. B. & A. Murugesan (2012) Use of Weather Information for Agricultural Decision Making. Weather, Climate, and Society, 5, 55-69.
- Garbrecht, J. D. & J. M. Schneider (2007) Climate forecast and prediction product dissemination for agriculture in the United States. *Australian Journal of Agricultural Research*, 58, 966-974.
- Garbrecht, J. D., J. M. Schneider & M. W. Van Liew (2006) Monthly runoff predictions based on rainfall forecasts in a small Oklahoma watershed. *Journal of the American Water Resources Association*, 42, 1285-1295.
- Garbrecht, J. D., X. C. Zhang, J. M. Schneider & J. L. Steiner (2010) Utility of Seasonal Climate Forecasts in Management of Winter-Wheat Grazing. *Applied Engineering in Agriculture*, 26, 855-866.
- George, D. A., C. J. Birch, J. F. Clewett, A. Wright, W. Allen & D. U. Keogh (2007) Needs for applied climate education in agriculture. Australian Journal of Experimental Agriculture, 47, 1-12.
- Gershunov, A. & T. P. Barnett (1998) ENSO influence on intraseasonal extreme rainfall and temperature frequencies in the contiguous United States: Observations and model results. *Journal of Climate*, 11, 1575-1586.
- Gleick, P. H. (2014) Water, Drought, Climate Change, and Conflict in Syria. Weather, Climate, and Society, 6, 331-340.
- Goddard, L., Y. Aitchellouche, W. Baethgen, M. Dettinger, R. Graham, P. Hayman, M. Kadi, R. Martínez & H. Meinke (2010) Providing Seasonal-to-Interannual Climate Information for Risk Management and Decision-making. *Procedia Environmental Sciences*, 1, 81-101.
- Goddard, L., J. W. Hurrell, B. P. Kirtman, J. Murphy, T. Stockdale & C. Vera (2012) Two Time Scales for the Price of One (almost). *BAMS*, 93, 621-629.
- Goddard, L., S. J. Mason, S. E. Zebiak, C. F. Ropelewski, R. Basher & M. A. Cane (2001) Current approaches to seasonal to internanual climate predictions. *International Journal of Climatology*, 21, 1111-1152.
- Groisman, P. Y., R. W. Knight, D. R. Easterling, T. R. Karl, G. C. Hegerl & V. N. Razuvaev (2005) Trends in Intense Precipitation in the Climate Record. *Journal of Climate*, 18, 1326-1350.
- Haigh, T., E. Takle, J. Andresen, M. Widhalm, J. S. Carlton & J. Angel (2015) Mapping the decision points and climate information use of agricultural producers across the U.S. Corn Belt. *Climate Risk Management*, 7, 20-30.
- Hale, J. 1991. The Best of The Old Farmer's Almanac. New York: Random House.
- Han, W., Z. Yang, L. Di & R. Mueller (2012) CropScape: A Web service based application for exploring and disseminating US conterminous geospatial cropland data products for decision support. *Computers and Electronics in Agriculture*, 84, 111-123.
- Hansen, J. W. (2002) Realizing the potential benefits of climate prediction to agriculture: issues, approaches, challenges. *Agricultural Systems*, 74, 309-330.
- Hansen, J. W., A. W. Hodges & J. W. Jones (1998) ENSO Influences on Agriculture in the Southeastern United States. *Journal of Climate*, 11, 404-411.
- Hansen, J. W. & M. Indeje (2004) Linking dynamic seasonal climate forecasts with crop simulation for maize yield prediction in semi-arid Kenya. *Agricultural and Forest Meteorology*, 125, 143-157.
- Hansen, J. W., J. W. Jones, A. Irmak & F. Royce. 2001. El Niño-Southern Oscillation Impacts on Crop Production in the Southeast United States. In *Impacts of El Niño and Climate Variability on Agriculture*, ed. C. Rosenzweig, 55-76. American Society of Agronomy, Crop Science Society of America, and Soil Science Society of America.
- Hartmann, H. C., R. Bales & S. Sorooshian (2002) Weather, climate, and hydrologic forecasting for the US Southwest: a survey. *Climate Research*, 21, 239-258.
- Hartmann, H. C., Pagano, T.C., Sorooshian, S., Bales, R. (2002) Confidence Builders Evaluating seasonal Climate Forecasts from User Perspectives. *BAMS*, 83, 683-698.
- Harwood, J., R. Heifner, K. Coble, J. Perry & A. Somwaru. 1999. Managing Risk in Farming: Concepts, Research, and Analysis., 130. Market and Trade Economics Division and Resource Economics Division.
- Hewitt, C., C. Buontempo & P. Newton (2013) Using Climate Predictions to Better Serve Society's Needs. Eos, Transactions American Geophysical Union, 94, 105-107.

- Higgins, R. W., Kousky, V.E., Xie, P. (2011) Extreme Precipitation Events in the South-Central United States during May and June 2010: Historical Perspective, Role of ENSO, and Trends. *Journal of Hydrometeorology*, 12, 1056-1070.
- Hill, H., M. Hadarits, R. Rieger, G. Strickert, E. G. R. Davies & K. M. Strobbe (2014) The Invitational Drought Tournament: What is it and why is it a useful tool for drought preparedness and adaptation? *Weather and Climate Extremes*, 3, 107-116.
- Hoskins, B. (2013) The potential for skill across the range of the seamless weather-climate prediction problem: a stimulus for our science. *Quarterly Journal of the Royal Meteorological Society*, 139, 573-584.
- Hu, Q., L. M. PytlikZillig, G. D. Lynne, A. J. Tomkins, W. J. Waltman, M. J. Hayes, K. G. Hubbard, I. Artikov, S. J. Hoffman & D. A. Wilhite (2006) Understanding Farmers' Forecast Use from Their Beliefs, Values, Social Norms, and Perceived Obstacles*. *Journal of Applied Meteorology and Climatology*, 45, 1190-1201.
- Huang, D. & S. Gao (2017) Impact of different cumulus convective parameterization schemes on the simulation of precipitation over China. *Tellus A: Dynamic Meteorology and Oceanography*, 69, 1406264.
- Hubbard, K. G. (2007) Agricultural Climatology. Journal of Service Climatology, 1, 1-9.
- Hudson, J. P. 1972. Agronomic implications of long-term weather forecasting. In Weather Forecasting for Agriculture and Industry. A Symposium., ed. J. A. Taylor, 44-55. Fairleigh Dickinson University Press.
- Infanti, J. M. & B. P. Kirtman (2014) Southeastern U.S. Rainfall Prediction in the North American Multi-Model Ensemble. *Journal of Hydrometeorology*, 15, 529-550.
- IPCC. 2013. Annex III: Glossary [Planton, S. (ed.)]. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, eds. T. F. Stocker, D. Qin, G.-K. Plattner, M. M. B. Tignor, S. K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex & P. M. Midgley. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Jia, L. 2017. Personal communication. ed. T. Klemm.
- Jia, L., G. A. Vecchi, X. Yang, R. G. Gedgel, T. L. Delworth, W. F. Stern, K. Paffendorf, S. D. Underwood & F. Zeng (2016) The Roles of Radiative Forcing, Sea Surface Temperatures, and Atmospheric and Land Initial Conditions in U.S. Summer Warming Episodes. *Journal of Climate*, 29, 4121-4135.
- Jia, L., X. Yang, G. A. Vecchi, R. G. Gudgel, T. L. Delworth, A. Rosati, W. F. Stern, A. T. Wittenberg, L. Krishnamurthy, S. Zhang, R. Msadek, S. Kapnick, S. Underwood, F. Zeng, W. G. Anderson, V. Balaji & K. Dixon (2015) Improved Seasonal Prediction of Temperature and Precipitation over Land in a High-Resolution GFDL Climate Model. *Journal of Climate*, 28, 2044-2062.
- Jin, E. K., J. L. Kinter, III, B. Wang, C. K. Park, I. S. Kang, B. P. Kirtman, J. S. Kug, A. Kumar, J. J. Luo, J. Schemm, J. Shukla & T. Yamagata (2008) Current status of ENSO prediction skill in coupled oceanatmosphere models. *Climate Dynamics*, 31, 647-664.
- Jones, J. W., J. W. Hansen, F. S. Royce & C. D. Messina (2000) Potential benefits of climate forecasting to agriculture. *Agriculture, Ecosystems & Environment*, 82, 169-184.
- Kansas State University. 1997. Wheat Production Handbook. Manhattan, Kansas: K-State Research & Extension.
- Kerr, R. A. (1989) A New Way to Forecast Next Season's Climate: A new mathematical technique that makes 90-day climate forecasts by searching the historic record does as well as humans. *Science*, 244, 30-31.
- --- (2008) Seasonal-Climate Forecasts Improving Ever So Slowly. Science, 321, 900-901.

Kirtman, B. P. & D. Min (2009) Multimodel Ensemble ENSO Prediction with CCSM and CFS. Monthly Weather Review, 137, 2908-2930.

- Kirtman, B. P., D. Min, J. M. Infanti, J. L. Kinter, D. A. Paolino, Q. Zhang, H. van den Dool, S. Saha, M. P. Mendez, E. Becker, P. Peng, P. Tripp, J. Huang, D. G. DeWitt, M. K. Tippett, A. G. Barnston, S. Li, A. Rosati, S. D. Schubert, M. Rienecker, M. Suarez, Z. E. Li, J. Marshak, Y.-K. Lim, J. Tribbia, K. Pegion, W. J. Merryfield, B. Denis & E. F. Wood (2014) The North American Multimodel Ensemble: Phase-1 Seasonal-to-Interannual Prediction; Phase-2 toward Developing Intraseasonal Prediction. *Bulletin of the American Meteorological Society*, 95, 585-601.
- Klemm, T. & R. A. McPherson (2017) The development of seasonal climate forecasting for agricultural producers. *Agricultural and Forest Meteorology*, 232, 384-399.
- --- (in review) What Farmers Need: Assessing Decision Timing and Seasonal Climate Forecast Needs of Winter Wheat Producers in the South-Central U.S. *Journal of Applied Meteorology and Climatology*.
- Klockow, K. E., R. A. McPherson & D. S. Sutter (2010) On the Economic Nature of Crop Production Decisions Using the Oklahoma Mesonet. *Weather, Climate, and Society*, 2, 224-236.
- Knutti, R., R. Furrer, C. Tebaldi, J. Cermak & G. A. Meehl (2010) Challenges in Combining Projections from Multiple Climate Models. *Journal of Climate*, 23, 2739-2758.
- Krakauer, N. Y., M. D. Grossberg, I. Gladkova & H. Aizenman (2013) Information Content of Seasonal Forecasts in a Changing Climate. Advances in Meteorology, 2013, 12.
- Lamb, P. J., R. P. Timmer & M. I. Lélé (2011) Professional development for providers of seasonal climate prediction. *Climate Research*, 47, 57-75.
- Latif, M., D. Anderson, T. Barnett, M. Cane, R. Kleeman, A. Leetmaa, J. O'Brien, A. Rosati & E. Schneider (1998) A review of the predictability and prediction of ENSO. *Journal of Geophysical Research: Oceans*, 103, 14375-14393.

- Lau, K.-M., K.-M. Kim & S. S. P. Shen (2002) Potential predictability of seasonal precipitation over the United States from canonical ensemble correlation predictions. *Geophysical Research Letters*, 29, 1-4.
- Lemos, M. C., C. J. Kirchhoff & V. Ramprasad (2012) Narrowing the climate information usability gap. Nature Climate Change, 2, 789.
- Lemos, M. C. & B. J. Morehouse (2005) The co-production of science and policy in integrated climate assessments. *Global Environmental Change*, 15, 57-68.
- Lemos, M. C. & R. B. Rood (2010) Climate projections and their impact on policy and practice. *Wiley Interdisciplinary Reviews: Climate Change*, 1, 670-682.
- Lorenz, E. N. (1969) The predictability of a flow which possesses many scales of motion. Tellus, 21, 289-307.
- Lyon, B., A. Giannini, P. Gonzalez & A. W. Robertson (2014) The role of targeted climate research at the IRI. *Earth Perspectives*, 1, 1-18.
- Maraun, D., F. Wetterhall, A. M. Ireson, R. E. Chandler, E. J. Kendon, M. Widmann, S. Brienen, H. W. Rust, T. Sauter, M. Themeßl, V. K. C. Venema, K. P. Chun, C. M. Goodess, R. G. Jones, C. Onof, M. Vrac & I. Thiele-Eich (2010) Precipitation downscaling under climate change: Recent developments to bridge the gap between dynamical models and the end user. *Reviews of Geophysics*, 48, n/a-n/a.
- Marthews, T. R., F. E. L. Otto, D. Mitchell, S. J. Dadson & R. G. Jones (2015) The 2014 Drought in the Horn of Africa: Attribution of Meteorological Drivers. Bulletin of the American Meteorological Society, 96, S83-S88.
- Mason, S. J., L. Goddard, N. E. Graham, E. Yulaeva, L. Sun & P. A. Arkin (1999) The IRI Seasonal Climate Prediction System and the 1997/98 El Niño Event. Bulletin of the American Meteorological Society, 80, 1853-1873.
- Mauget, S., J. Zhang & J. Ko (2009) The Value of ENSO Forecast Information to Dual-Purpose Winter Wheat Production in the U.S. Southern High Plains. *Journal of Applied Meteorology and Climatology*, 48, 2100-2117.
- Mavi, H. S. & G. J. Tupper. 2004. Agrometeorology Principles and Applications of Climate Studies in Agriculture. Food Products Press.
- McPherson, R. A., C. A. Fiebrich, K. C. Crawford, J. R. Kilby, D. L. Grimsley, J. E. Martinez, J. B. Basara, B. G. Illston, D. A. Morris, K. A. Kloesel, A. D. Melvin, H. Shrivastava, J. M. Wolfinbarger, J. P. Bostic, D. B. Demko, R. L. Elliott, S. J. Stadler, J. D. Carlson & A. J. Sutherland (2007) Statewide Monitoring of the Mesoscale Environment: A Technical Update on the Oklahoma Mesonet. *Journal of Atmospheric and Oceanic Technology*, 24, 301-321.
- Meadow, A. M., D. B. Ferguson, Z. Guido, A. Horangic, G. Owen & T. Wall (2015) Moving toward the Deliberate Coproduction of Climate Science Knowledge. *Weather, Climate, and Society,* **7,** 179-191.
- Meinke, H., S. M. Howden, W. Baethgen, G. L. Hammer, R. Selvaraju & R. Stone. 2003. Can Climate Knowledge Lead to Better Rural Policies and Risk Management Practices? In NOAA workshop: "Insights and Tools for Adaptation: Learning from Climate Variability", Washington DC, November 2003. Paper from the NOAA workshop "Insights and Tools for Adaptation: Learning from Climate Variability", Washington, D.C., November 2003.
- Meinke, H. & R. C. Stone (2005) Seasonal and Inter-Annual Climate Forecasting: The New Tool for Increased Preparedness to Climate Variability and Change in Agricultural Planning and Operations. *Climatic Change*, 70, 221-253.
- Miles, E. L., A. K. Snover, L. C. W. Binder, E. S. Sarachik, P. W. Mote & N. Mantua (2006) An approach to designing a national climate service. *Proceedings of the National Academy of Sciences of the United States of America*, 103, 19616-19623.
- Mishra, A., J. W. Hansen, M. Dingkuhn, C. Baron, S. B. Traoré, O. Ndiaye & M. N. Ward (2008) Sorghum yield prediction from seasonal rainfall forecasts in Burkina Faso. *Agricultural and Forest Meteorology*, 148, 1798-1814.
- Moser, S. C. (2010) Now more than ever: The need for more societally relevant research on vulnerability and adaptation to climate change. *Applied Geography*, 30, 464-474.
- Mulholland, D. P., P. Laloyaux, K. Haines & M. A. Balmaseda (2015) Origin and Impact of Initialization Shocks in Coupled Atmosphere–Ocean Forecasts. *Monthly Weather Review*, 143, 4631-4644.
- Muller, W. A., C. Appenzeller & C. Schar (2005) Probabilistic seasonal prediction of the winter North Atlantic Oscillation and its impact on near surface temperature. *Climate Dynamics*, 24, 213-226.
- Musser, W. N. & G. F. Patrick. 2002. How Much does Risk Really Matter to Farmers? In A Comprehensive Assessment of the Role of Risk in U.S. Agriculture, eds. R. E. Just & R. D. Pope, 537-556. Springer US.
- Mutiibwa, D., S. J. Vavrus, S. A. McAfee & T. P. Albright (2015) Recent spatiotemporal patterns in temperature extremes across conterminous United States. *Journal of Geophysical Research-Atmospheres*, 120, 7378-7392.
- Nadeau, R. & R. G. Niemi (1995) Educated Guesses: The Process of Answering Factual Knowledge Questions in Surveys. Public Opinion Quarterly, 59, 323-346.
- Namias, J. (1968) Long-range weather forecasting—History, current status and outlook. Bulletin of the American Meteorological Society, 49, 438-470.
- Nicholls, N. (1980) Long-range weather forecasting: Value, status, and prospects. Reviews of Geophysics, 18, 771-788.
- NOAA. 2011. 9th Annual Climate Prediction Applications Science Workshop report. In NOAA Report, 32.
- North, R., M. Trueman, M. Mittermaier & M. J. Rodwell (2013) An assessment of the SEEPS and SEDI metrics for the verification of 6h forecast precipitation accumulations. *Meteorological Applications*, 20, 164-175.

- NRC. 2010. Assessment of Intraseasonal to Interannual Climate Prediction and Predictability. Washington, DC: The National Academies Press.
- NRDC. 2013. Record-Breaking \$17.3 Billion in Crop Losses Last Year; Significant Portion Potentially Avoidable.
- Nulty, D. D. (2008) The adequacy of response rates to online and paper surveys: what can be done? Assessment & Evaluation in Higher Education, 33, 301-314.
- O'Lenic, E. A., D. A. Unger, M. S. Halpert & K. S. Pelman (2008) Developments in Operational Long-Range Climate Prediction at CPC (with corrigendum attached). *Weather and Forecasting*, 23, 496-515.
- Palmer, T. N., A. Alessandri, U. Andersen, P. Cantelaube, M. Davey, P. Delecluse, M. Deque, E. Diez, F. J. Doblas-Reyes, H. Feddersen, R. Graham, S. Gualdi, J. F. Gueremy, R. Hagedorn, M. Hoshen, N. Keenlyside, M. Latif, A. Lazar, E. Maisonnave, V. Marletto, A. P. Morse, B. Orfila, P. Rogel, J. M. Terres & M. C. Thomson (2004) Development of a European multimodel ensemble system for seasonal-to-interannual prediction (DEMETER). *Bulletin of the American Meteorological Society*, 85, 853-+.
- Pasteris, P. A., T. Puterbaugh & R. Motha. 2004. Climate Services A USDA Perspective. In 84th American Meteorological Society Annual Meeting, 6 p.
- Peng, P., A. G. Barnston & A. Kumar (2013) A Comparison of Skill between Two Versions of the NCEP Climate Forecast System (CFS) and CPC's Operational Short-Lead Seasonal Outlooks. *Weather and Forecasting*, 28, 445-462.
- Peng, P., A. Kumar, M. S. Halpert & A. G. Barnston (2012) An Analysis of CPC's Operational 0.5-Month Lead Seasonal Outlooks. *Weather and Forecasting*, 27, 898-917.
- Peppler, R. A. (2010) "Old Indian Ways" of Predicting the Weather: Senator Robert S. Kerr and the Winter Predictions of 1950–51 and 1951–52. Weather, Climate, and Society, 2, 200-209.
- Polade, S. D., A. Gershunov, D. R. Cayan, M. D. Dettinger & D. W. Pierce (2013) Natural climate variability and teleconnections to precipitation over the Pacific-North American region in CMIP3 and CMIP5 models. *Geophysical Research Letters*, 40, 2296-2301.
- Power, S. B., N. Plummer & P. Alford (2007) Making climate model forecasts more useful. Australian Journal of Agricultural Research, 58, 945-951.
- PRISM Climate Group. Oregon State University, .
- Prokopy, L. S., T. Haigh, A. S. Mase, J. Angel, C. Hart, C. Knutson, M. C. Lemos, Y.-J. Lo, J. McGuire, L. W. Morton, J. Perron, D. Todey & M. Widhalm (2013) Agricultural Advisors: A Receptive Audience for Weather and Climate Information? *Weather, Climate, and Society*, 5, 162-167.
- Ramirez-Rodrigues, M. A., S. Asseng, C. W. Fraisse, L. Stefanova & A. Eisenkolbi (2014) Tailoring wheat management to ENSO phases for increased wheat production in Paraguay. *Climate Risk Management*, 3, 24-38.
- Rasmusson, E. M. & T. H. Carpenter (1982) Variations in Tropical Sea Surface Temperature and Surface Wind Fields Associated with the Southern Oscillation/El Niño. *Monthly Weather Review*, 110, 354-384.
- Richardson, D. S. (2001) Measures of skill and value of ensemble prediction systems, their interrelationship and the effect of ensemble size. *Quarterly Journal of the Royal Meteorological Society*, 127, 2473-2489.
- Ringler, T., L. Ju & M. Gunzburger (2008) A multiresolution method for climate system modeling: application of spherical centroidal Voronoi tessellations. *Ocean Dynamics*, 58, 475-498.
- Robertson, A. W., A. Kumar, M. Peña & F. Vitart (2015) Improving and Promoting Subseasonal to Seasonal Prediction. Bulletin of the American Meteorological Society, 96, ES49-ES53.
- Rodwell, M. J., D. S. Richardson, T. D. Hewson & T. Haiden (2010) A new equitable score suitable for verifying precipitation in numerical weather prediction. *Quarterly Journal of the Royal Meteorological Society*, 136, 1344-1363.
- Roel, A. & W. E. Baethegan. 2007. Towards the Development of a Spatial Decision Support System (SDSS) for the Application of Climate Forecasts in Uruguayan Rice Production Sector. In *Climate Prediction and Agriculture*, eds. J. W. Hansen & M. V. K. Sivakumar, 89-97. Springer.
- Roncoli, C., C. Jost, P. Kirshen, M. Sanon, K. Ingram, M. Woodin, L. Somé, F. Ouattara, B. Sanfo, C. Sia, P. Yaka & G. Hoogenboom (2009) From accessing to assessing forecasts: an end-to-end study of participatory climate forecast dissemination in Burkina Faso (West Africa). *Climatic Change*, 92, 433-460.
- Ropelewski, C. F. & M. S. Halpert (1986) North American Precipitation and Temperature Patterns Associated with the El Niño/Southern Oscillation (ENSO). *Monthly Weather Review*, 114, 2352-2362.
- Saha, S., S. Moorthi, X. Wu, J. Wang, S. Nadiga, P. Tripp, D. Behringer, Y.-T. Hou, H.-y. Chuang, M. Iredell, M. Ek, J. Meng, R. Yang, M. P. Mendez, H. van den Dool, Q. Zhang, W. Wang, M. Chen & E. Becker (2014) The NCEP Climate Forecast System Version 2. *Journal of Climate*, 27, 2185-2208.
- Schneider, J. M. & J. D. Garbrecht (2003a) A measure of the usefulness of seasonal precipitation forecasts for agricultural applications. *Transactions of the Asae*, 46, 257-267.

Schneider, J. M. & J. D. Garbrecht. 2003b. Regional Utility of NOAA/CPC Seasonal Climate Precipitation Forecasts. In World Water & Environmental Resources Congress 2003, 1-8.

Schneider, J. M. & J. D. Garbrecht (2006) Dependability and effectiveness of seasonal forecasts for agricultural applications. *Transactions of the Asabe*, 49, 1737-1753.

- Schneider, J. M. & J. D. Wiener (2009) Progress toward filling the weather and climate forecast need of agricultural and natural resource management. *Journal of Soil and Water Conservation*, 64, 100A-106A.
- Schoof, J. T. & S. M. Robeson (2016) Projecting changes in regional temperature and precipitation extremes in the United States. *Weather and Climate Extremes*, 11, 28-40.
- Shafiee-Jood, M., X. Cai, L. Chen, X.-Z. Liang & P. Kumar (2014) Assessing the value of seasonal climate forecast information through an end-to-end forecasting framework: Application to U.S. 2012 drought in central Illinois. *Water Resources Research*, 50, 6592-6609.
- Slingo, J. & T. Palmer (2011) Uncertainty in weather and climate prediction. Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences, 369, 4751-4767.
- Sobie, S. R. & T. Q. Murdock (2017) High-Resolution Statistical Downscaling in Southwestern British Columbia. *Journal of Applied Meteorology and Climatology*, 56, 1625-1641.
- Sonka, S. T., S. L. Hofing & S. A. Changnon (1992) How Agribusiness Uses Climate Predictions: Implications for Climate Research and Provision of Predictions. *Bulletin of the American Meteorological Society*, 73, 1999-2008.
- Southern Climate Impacts Planning Program. 2018. Average monthly temperature and precipitation tool.
- Steinemann, A. C. (2006) Using Climate Forecasts for Drought Management. Journal of Applied Meteorology and Climatology, 45, 1353-1361.
- Steiner, J. L., J. M. Schneider, J. D. Garbrecht & X. J. Zhang. 2004. Climate Forecasts: Emerging Potential to Reduce Dryland Farmers' Risks. In *Challenges and Strategies of Dryland Agriculture*, eds. S. C. Rao & J. Ryan, 47-65. Madison, WI: Crop Science Society of America and American Society of Agronomy.
- Stern, P. C. & W. E. Easterling. 1999. Making Climate Forecasts Matter.
- Stockdale, T. N., O. Alves, G. Boer, M. Deque, Y. Ding, A. Kumar, K. Kumar, W. Landman, S. Mason, P. Nobre, A. Scaife, O. Tomoaki & W. T. Yun (2010) Understanding and Predicting Seasonal-to-Interannual Climate Variability - The Producer Perspective. *Procedia Environmental Sciences*, 1, 55-80.
- Svoboda, M., D. LeComte, M. Hayes, R. Heim, K. Gleason, J. Angel, B. Rippey, R. Tinker, M. Palecki, D. Stooksbury, D. Miskus & S. Stephens (2002) The Drought Monitor. *Bulletin of the American Meteorological Society*, 83, 1181-1190.
- Takle, E. S., C. J. Anderson, J. Andresen, J. Angel, R. W. Elmore, B. M. Gramig, P. Guinan, S. Hilberg, D. Kluck, R. Massey, D. Niyogi, J. M. Schneider, M. D. Shulski, D. Todey & M. Widhalm (2014) Climate Forecasts for Corn Producer Decision Making. *Earth Interactions*, 18, 1-8.
- Tall, A. (2013) What do we mean by Climate Services? WMO Bulletin, 62, 7-10.
- Taylor, A. L., S. Dessai & W. B. de Bruin (2015) Communicating uncertainty in seasonal and interannual climate forecasts in Europe. *Philosophical Transactions of the Royal Society of London A: Mathematical, Physical and Engineering Sciences*, 373.
- Templeton, S. R., M. Shane Perkins, H. D. Aldridge, W. C. J. Bridges & B. R. Lassiter (2014) Usefulness and uses of climate forecasts for agricultural extension in South Carolina, USA. *Regional Environmental Change*, 14, 645-655.
- Toth, Z., M. Peña & A. Vintzileos (2007) Bridging the Gap between Weather and Climate Forecasting: Research Priorities for Intraseasonal Prediction. *Bulletin of the American Meteorological Society*, 88, 1427-1429.
- U.S. Department of Agriculture. 2017. Summary of Business Reports and Data.
- Unganai, L. S., J. Troni, D. Manatsa & D. Mukarakate (2013) Tailoring seasonal climate forecasts for climate risk management in rainfed farming systems of southeast Zimbabwe. *Climate and Development*, **5**, 139-152.
- USDA. 2012. Ag Census.
- ---. 2015a. Coorporative Extension.
- ---. 2015b. Farm Computer Usage and Ownership.
- ---. 2015c. National Institute for Food and Agriculture. About NIFA.
- ---. 2016a. Program Statistics Summary of Premiums and Losses.
- ---. 2016b. Summary of Business Reports and Data.
- USDA ERS. 2013. U.S. Drought 2012: Farm and Food Impacts. http://www.ers.usda.gov/topics/in-the-news/usdrought-2012-farm-and-food-impacts.aspx#farms.
- van den Dool, H. M. 1994. New Operational Long-Lead Seasonal Climate Outlooks Out to One Year: Rationale. In Nineteenth Annual Climate Predictions Workshop, 405-407. College Park, MD.
- Vecchi, G. A., T. Delworth, R. Gudgel, S. Kapnick, A. Rosati, A. T. Wittenberg, F. Zeng, W. Anderson, V. Balaji, K. Dixon, L. Jia, H. S. Kim, L. Krishnamurthy, R. Msadek, W. F. Stern, S. D. Underwood, G. Villarini, X. Yang & S. Zhang (2014) On the Seasonal Forecasting of Regional Tropical Cyclone Activity. *Journal of Climate*, 27, 7994-8016.
- Villarini, G., J. A. Smith & G. A. Vecchi (2013) Changing Frequency of Heavy Rainfall over the Central United States. *Journal of Climate*, 26, 351-357.
- Vincelli, P., G. Parker & S. McNeill. 1995. Aflatoxins in Corn. ed. C. o. A. University of Kentucky.
- Wall, T. U., E. McNie & G. M. Garfin (2017) Use-inspired science: making science usable by and useful to decision makers. Frontiers in Ecology and the Environment, 15, 551-559.
- Weisheimer, A., F. J. Doblas-Reyes, T. N. Palmer, A. Alessandri, A. Arribas, M. Déqué, N. Keenlyside, M. MacVean, A. Navarra & P. Rogel (2009) ENSEMBLES: A new multi-model ensemble for seasonal-to-

annual predictions—Skill and progress beyond DEMETER in forecasting tropical Pacific SSTs. *Geophysical Research Letters*, 36, n/a-n/a.

- Wen, C., Y. Xue & A. Kumar (2012) Seasonal Prediction of North Pacific SSTs and PDO in the NCEP CFS Hindcasts. *Journal of Climate*, 25, 5689-5710.
- Werner, A. T. & A. J. Cannon (2016) Hydrologic extremes an intercomparison of multiple gridded statistical downscaling methods. *Hydrology and Earth System Sciences*, 20, 1483-1508.
- Wilks, D. S. 2011. Statistical methods in the atmospheric sciences. Amsterdam ; Boston: Elsevier/Academic Press.
- Wilmer, H. & M. E. Fernández-Giménez (2015) Rethinking rancher decision-making: a grounded theory of ranching approaches to drought and succession management. *The Rangeland Journal*, 37, 517-528.
- Wilmer, H., E. York, W. K. Kelley & M. W. Brunson (2016) "In Every Rancher's Mind": Effects of Drought on Ranch Planning and Practice. *Rangelands*, 38, 216-221.
- WMO. 2015. Global Producing Centres for Long Range Forecasting.
- Xie, P., M. Chen, S. Yang, A. Yatagai, T. Hayasaka, Y. Fukushima & C. Liu (2007) A Gauge-Based Analysis of Daily Precipitation over East Asia. *Journal of Hydrometeorology*, 8, 607-626.
- Zängl, G., D. Reinert, P. Rípodas & M. Baldauf (2014) The ICON (ICOsahedral Non-hydrostatic) modelling framework of DWD and MPI-M: Description of the non-hydrostatic dynamical core. *Quarterly Journal of the Royal Meteorological Society*, 141, 563-579.
- Zhai, P., X. Zhang, H. Wan & X. Pan (2005) Trends in Total Precipitation and Frequency of Daily Precipitation Extremes over China. *Journal of Climate*, 18, 1096-1108.
- Zhou, C. L. & K. C. Wang (2017) Contrasting Daytime and Nighttime Precipitation Variability between Observations and Eight Reanalysis Products from 1979 to 2014 in China. *Journal of Climate*, 30, 6443-6464.
- Zinyengere, N., T. Mhizha, E. Mashonjowa, B. Chipindu, S. Geerts & D. Raes (2011) Using seasonal climate forecasts to improve maize production decision support in Zimbabwe. *Agricultural and Forest Meteorology*, 151, 1792-1799.